Check for updates

#### **OPEN ACCESS**

EDITED BY Vincenzo Neri, University of Foggia, Italy

REVIEWED BY Tamer Alnaimy, Zagazig University, Egypt Michail Klimovskij, East Sussex Healthcare NHS Trust, United Kingdom

\*CORRESPONDENCE

Zi Ye ⊠ yezi1022@gmail.com

<sup>†</sup>These authors have contributed equally to this work

RECEIVED 08 January 2025 ACCEPTED 17 March 2025 PUBLISHED 14 April 2025

#### CITATION

Zhou R, Wang D, Zhang H, Zhu Y, Zhang L, Chen T, Liao W and Ye Z (2025) Vision techniques for anatomical structures in laparoscopic surgery: a comprehensive review.

Front. Surg. 12:1557153. doi: 10.3389/fsurg.2025.1557153

#### COPYRIGHT

© 2025 Zhou, Wang, Zhang, Zhu, Zhang, Chen, Liao and Ye. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

# Vision techniques for anatomical structures in laparoscopic surgery: a comprehensive review

Ru Zhou<sup>1†</sup>, Dan Wang<sup>2†</sup>, Hanwei Zhang<sup>3</sup>, Ying Zhu<sup>2</sup>, Lijun Zhang<sup>4</sup>, Tianxiang Chen<sup>5</sup>, Wengiang Liao<sup>1</sup> and Zi Ye<sup>3\*</sup>

<sup>1</sup>Department of General Surgery, RuiJin Hospital LuWan Branch, Shanghai Jiaotong University School of Medicine, Shanghai, China, <sup>2</sup>Hangzhou Institute for Advanced Study, University of Chinese Academy of Sciences, Zhejiang, Hangzhou, China, <sup>3</sup>Institute of Intelligent Software, Guangzhou, Guangdong, China, <sup>4</sup>Institute of Software Chinese Academy of Sciences, Beijing, China, <sup>5</sup>School of Cyber Space and Technology, University of Science and Technology of China, Hefei, China

Laparoscopic surgery is the method of choice for numerous surgical procedures, while it confronts a lot of challenges. Computer vision exerts a vital role in addressing these challenges and has become a research hotspot, especially in the classification, segmentation, and target detection of abdominal anatomical structures. This study presents a comprehensive review of the last decade of research in this area. At first, a categorized overview of the core subtasks is presented regarding their relevance and applicability to real-world medical scenarios. Second, the dataset used in the experimental validation is statistically analyzed. Subsequently, the technical approaches and trends of classification, segmentation, and target detection tasks are explored in detail, highlighting their advantages, limitations, and practical implications. Additionally, evaluation methods for the three types of tasks are discussed. Finally, gaps in current research are identified. Meanwhile, the great potential for development in this area is emphasized.

#### KEYWORDS

computer vision, laparoscopic surgery, segmentation, classification, object detection

# **1** Introduction

Laparoscopic surgery refers to an advanced, minimally invasive surgical technique. However, laparoscopic surgery faces many challenges, including limited field of view and image quality issues (1-3). This also raises the bar for surgeons, requiring enhanced spatial positioning, hand-eye coordination, and technical skills (4, 5). Computer vision (CV) has emerged as a promising tool to address these challenges, which can potentially enhance the accuracy and efficiency of laparoscopic procedures.

CV has made significant progress in its application to laparoscopic surgery through simulating the human visual system and utilizing algorithms such as image processing, machine learning, and deep learning(DL) to extract useful features from surgical images. It deals with a variety of problems associated with laparoscopic surgery (6), including improving image quality, providing real-time feedback, error detection during surgery, assisting in identifying and localizing anatomical structures. Artificial intelligence (AI)driven approaches have been developed for addressing these critical tasks. For example, machine learning models have been employed to classify surgical maneuvers, identify surgical stages, and even predict potential complications, therefore greatly aiding in surgical planning and execution (7). These innovations have improved the accuracy and efficiency of surgery, contributed to better training and assessment of surgical skills, as well as enhanced standardization and safety in surgery. Visual tasks concentrating on the organs in the abdomen can help surgeons by guiding them visually and assisting them to find and identify structures within the body (8). For example, accurately segmenting abdominal organs to show their contours and locations can contribute to avoiding damaging vital structures during surgery, reducing surgical risks, and improving the safety and success rate of surgery (9). Automatic identification and localization of lesion areas, foreign objects, or abnormal structures can provide doctors with diagnostic basis and treatment recommendations. This can facilitate rapid diagnosis, timely and effective treatment measures, shorten operation time, and improve the success rate of surgery and the patient's treatment experience.

Despite the advancements and the promising potential of CV in laparoscopic surgery, there is an increasing need for a comprehensive review to synthesize the existing research and guide future developments. The field is rapidly evolving, with numerous studies investigating various perspectives of CV applications in laparoscopic surgery. However, these studies usually focus on specific tasks or techniques, making it challenging for researchers and practitioners to have a clear and cohesive understanding of the overall landscape (10). A systematic review that collates and critically evaluates the current state of CV methods for anatomical structure analysis in laparoscopic images can provide valuable insights, identify existing literature gaps, and suggest future research directions. The review can also serve as a resource for developing standardized protocols and benchmarks for the evaluation of AI systems in this domain.

Currently, the segmentation, classification, and object detection tasks for abdominal organs in laparoscopy are intensively studied.

We searched for references on Google Scholar. The results of the search query revealed 117 papers associated with the topic. Specific data on the sources of articles from conferences, journals, and preprints, as well as other platforms, are shown in Figure 1. The time span is from 2014 to June 2024, with the majority coming from 2023 and 2024. We specifically examined articles from IEEE, Science Direct, PubMed, and Springer, as well as the proceedings of medical imaging conferences, including MICCAI, IPMI, ISBI, RSNA, and SPIE. Figure 5a illustrates the percentage of all included studies for the three tasks of segmentation, classification, and target detection, which we will examine in detail below and suggest future research directions. Therefore, this work has the following three contributions:

- Comprehensive Survey: We conducted an exhaustive survey of the existing literature on using CV in laparoscopic surgery, systematically searching and analyzing articles on segmentation, classification, and object detection of anatomical structures.
- Task-specific categorization: We systematically categorize the core subtasks in the context of real-world healthcare scenarios, emphasizing their relevance and applicability in the clinical environment.
- Data and Metrics: We discuss the datasets commonly applied in this field, their characteristics, and the importance of using consistent and comprehensive evaluation indicators in order to facilitate fair comparison and evaluation of methods.
- Methodological Insights: We discuss the reviewed literature from task-specific and methodological perspectives, emphasizing their strengths, limitations, as well as practical implications.



The number of papers searched on other platforms, such as conferences, different types of journals, preprint platforms, etc., according to the selection criteria, are listed by different years, ranging from 2014 to June 2024.

• Future Directions: We identify key challenges, indicate potential research directions, and highlight the potential for future developments to advance CV use in laparoscopic surgery.

This study is structured as follows: Section 2 describes the specific healthcare scenarios and tasks covered in the selected literature. Section 3 describes in detail the datasets used in the relevant publications. Then, Section 4 provides a detailed analysis of the methods in the related literature from different perspectives, including learning strategies. Finally, Section 5 summarizes the results of this study, highlights the main findings, and provides an outlook on possible further developments in this research area.

# 2 Core tasks and application scenarios

As presented in Figure 2, seven types of core tasks are summarized in the study for classification, segmentation, and target detection of anatomical structures in laparoscopic surgical images. These sub-tasks and their application scenarios will be detailed in the following section in order to investigate their importance and application effects in the actual surgical process.

## 2.1 Classification of anomalous regions

This task is typically a binary classification task helping physicians to make faster and more accurate diagnoses and treatment plans by classifying images as normal or abnormal, as presented in Figure 2a (11). For an image I, find f such that:

$$f(I) = \{\text{normal, anomalous}\}$$
(1)

In Equation 1, normal represents the normal category and anomalous represents the abnormal category.

Common subtasks include classification of endometriosis, which is difficult because the differences between normal and pathological tissues are difficult to discern for non-specialists, the



lesion sites have diversity and complexity, and there exist significant variations in visual appearance within and between people (12). Image classification techniques can significantly enhance the accuracy of diagnosing endometriosis and reduce the need for multiple procedures due to missed lesion sites. Visalaxi et al. (13) applied Resnet50 to automatically diagnose endometriosis with 91% accuracy.

## 2.2 Classification of anatomical structures

This task assigns category labels to an entire image or region of interest (ROI). Common tasks include organ categorization, i.e., categorizing different organs in an image or rating a class of organs, as presented in Figure 2b (14). Considering an image or ROI R, find a function f such that:

$$f(R) = l \tag{2}$$

In Equation 2,  $l \in \mathbb{L}_{\text{structures}}$  and  $\mathbb{L}_{\text{structure}}$  represents the type of anatomical structures.

For example, full-resolution convolutional neural network (FrCNN) (15) is adopted for automated detection of different organs such as the uterus, ovaries, fallopian tubes. Another challenging task is grading the vascularity of the gallbladder wall, which involves scrutinizing the vascular morphology as well as the thickness, inflammatory state, and fat coverage of the gallbladder wall. Fatty infiltration or thickening of the gallbladder wall adds to the difficulty of evaluation (16). Loukas et al. (17) achieved image- and video-level classification of blood vessels in the gallbladder wall through multiple instance learning.

# 2.3 Semantic segmentation of anatomical structures

The task is aimed at assigning a category label to each pixel in an image containing common abdominal anatomical structures such as liver, kidney, gallbladder, and fat. This segmentation method goes beyond simply dividing up regions in an image to comprehend the semantic information of the various regions, i.e., the type of anatomical structure that each region represents, as presented in Figure 2c (18). Given an input image *I*, the objective is to find a function *f* such that:

$$f(I) = \{L_{ij}\}_{i=1,j=1}^{H,W}$$
(3)

In Equation 3,  $L_{ij} \in \mathbb{L}_{\text{structure}}$  and  $\mathbb{L}_{\text{structure}}$  is the set of class labels for anatomical structures,  $H \times W$  represent the dimensions of the image.

Transanal total rectal mesorectal excision (TaTME) has become the focus of recent laparoscopic research and is an important tool for treating colon and rectal cancers. LDCNet (19) can segment organs and tissues appearing in TaTME, such as the liver, gallbladder, spleen, and intestines. So that robots could help with anterior rectal resections or rectal extirpations, Kolbinger et al. (20) made a machine-learning model that can correctly name 11 types of tissues.

Compared to larger organs, tissues are representatively more dispersed, smaller, finer in granularity, and highly variable in appearance. They may blend with the surrounding anatomical region, making precise segmentation more challenging. Loose connective tissue fibers (LCTFs) are a complicated, hard-to-spot anatomical structure. Careful removal of loose connective tissue fibers improves cancer prognosis and reduces the risk of surgical complications. Kumazu et al. (21) then trained Unet to segment LCTFs. Through cutting away loose connective tissue in the rectal mesentery, Total Mesorectal Excision(TME) lowers the risk of positive circumferential margins and is a strong indicator of local recurrence (22). SwinPA-Net (23) identifies loose connective tissue in the rectal mesentery, helping to avoid damage to vital tissues such as blood vessels and nerves during surgery.

In addition, artificially defined anatomical regions are needed in specific surgical scenarios, usually areas that need to be precisely manipulated or observed during surgery. The surgeon must accurately remove the mesenteric tissue surrounding the rectum during TME. As a result, a common semantic segmentation task is identifying and labeling the anatomical lines of the rectal mesentery (24).

## 2.4 Detection of lesion

The current task aims to identify and localize possible lesion regions in an image, determining the presence of a lesion and its location and labeling it with a bounding box, as presented in Figure 2d (18). Given an image I, find f such that:

$$f(I) = \{B_i\}_{i=1}^M$$
(4)

In Equation 4,  $B_i$  are the coordinates of the lesion,  $B_i = (x_{i1}, y_{i1}, x_{i2}, y_{i2})$ 

Leibetseder et al. (18) applied Faster R-CNN and Mask R-CNN to find areas of endometriosis in laparoscopic gynecological videos and give confidence to those areas.

#### 2.5 Detection of anatomical structures

Typically, lesion identification tasks detect relatively small and fewer targets because lesions are usually localized anomalous regions in an image. By contrast, the anatomical structure recognition task can be more complex. The task aims to automatically identify and localize different anatomical structures appearing in an image, usually involving multiple targets of different shapes and sizes, each with its own unique characteristics, as presented in Figure 2e (25, 26). For an image *I*, find *f* such that:

$$f(I) = \{(B_i, l_i)\}_{i=1}^N$$
(5)

In Equation 5,  $B_i$  are the bounding boxes,  $B_i = (x_{i1}, y_{i1}, x_{i2}, y_{i2})$ and  $l_i$  represent the anatomical structures labels.

Boonkong et al. (25) employed DNNs to detect the uterus in laparoscopic images. Cui et al. (27) introduced a YOLOv4-based method for recognizing vas deferens images in laparoscopic inguinal hernia repair surgery. Moreover, specific critical points are common targets for detection. They may not be actual anatomical structures, but rather "judgments" made by the surgeon based on laparoscopic images. Detecting these anatomical landmarks helps the surgeon locate organ tissues and assess their morphology, location, and interrelationships (26).

## 2.6 Recognition of surgical action triplet

In 2014, Katic et al. (28) developed the surgical action triplet: linstrument, verb, target¿. This is a task that involves correctly identifying surgical instruments, actions that are being performed, and body parts that they make interactions within complicated laparoscopic videos, as presented in Figure 2f (29). The difficulty lies in the surgical instruments' nuances, the actions' temporal properties, and the similarity of the target organs. To be specific, identifying surgical instruments requires a detailed analysis of their heads and handles. The interaction between the instrument and the target organ needs to be taken into consideration. Secondly, the color and texture of intraabdominal organs may be similar due to the influence of fat or connective tissue, further increasing the difficulty of identification. In addition, the identification of the triad is highly associated with the temporal information in the video. Several of the above issues lead to making recognizing surgical triplets a challenging task. CholecTriplet2021 (30) and CholecTriplet2022 challenge (31) are endoscopic vision challenges organized by MICCAI to identify surgical action triplets in laparoscopic videos.

Given an image sequence  $\{I_t\}_{t=1}^T$ , find f such that:

$$f\left(\{I_t\}_{t=1}^T\right) = (\mathbb{I}, \mathbb{A}, \mathbb{T})$$
(6)

In Equation 6,  $\mathbb{I}$  is the set of instruments,  $\mathbb{A}$  refers to the set of actions, and  $\mathbb{T}$  indicates the set of targets.

## 2.7 Composite task

In addition to the above-mentioned tasks involving only segmentation, classification, or detection, other tasks may require multiple stages, usually combining the steps of segmentation, classification, and detection, such as predicting a critical view of safety (CVS) in laparoscopic cholecystectomy (LC), as presented in Figure 2g (32).

In LC, CVS is often used as a standard operating procedure (33). CVS is the most important field to confirm the safety of the operation, and it can only be achieved if the three conditions of "access to the gallbladder by the cystic duct and the cystic artery

only" are met simultaneously (34). Therefore, predicting the CVS usually involves a two-step process: (1) accurately identifying and locating the critical tissues. localization of key tissues. (2) Reasoning about the geometric relationships between the tissues and determining the quality of their exposure to determine whether the CVS criteria are met.

The TCNN model (35) is a model that segments the hepatic capsule structures first and then evaluates the CVS using the segmentation masks. In contrast to TCNN, the Murali et al. (36) trained using only bounding box annotations, outperforming several benchmark methods and scales efficiently when trained based on segmentation masks. In addition to semantic segmentation or target detection as an intermediate step, Alapatt et al. (37) also proposed a direct prediction of CVS based on self-supervised learning without prior segmentation or identification of gallbladder structures.

# 3 Public datasets

Recently, freely available datasets have exerted a central impact on developing new methods for segmentation, classification, and target detection of abdominal organs and tissues from laparoscopic images.

With the consideration of 117 publications, it was found that totally over 90 datasets were used, categorized as public datasets, private datasets, and non-conforming datasets. "Conforming" is defined as being related to the tasks studied in this paper. Some studies involve multiple tasks, including surgical stage identification, surgical instrument segmentation, etc. The datasets which could be adopted for these additional tasks were not considered to meet the criteria. As shown in Table 1, it should be noted that the "Application" column refers to the types of tasks that can be applied to the dataset. Specifically, some of the datasets in the list are designed for such tasks, which we call specialized datasets, and the rest are generalized datasets. Figure 3 illustrates the number of times all publicly available datasets are used, as well as the percentage of generalized and specialized datasets.

We have selected six of the most extensively used public datasets and categorized them into generic and specialized, which are explored separately in the following subsections.

## 3.1 General datasets

Generic datasets often lack specific labeling or annotations for the visual tasks studied in this paper, therefore requiring additional annotation work. These datasets are invaluable due to their large number and variety of images, and provide a wide base for all types of research and development of AI applications in laparoscopic surgery.

#### 3.1.1 Cholec80

The Cholec80 dataset (38) contains 80 LC videos. Each of these frames is labeled with a surgical stage and tool presence, which can

#### TABLE 1 Public datasets.

Dataset	Year	Size	Procedure <sup>a</sup>	Applications	References
Cholec80	2016	80 videos	LC	Surgical phase recognition, Surgical instrument presence detection	(15, 37–40, 43, 50, 55, 84, 96–99)
M2cai16-workflow	2016	41 videos	LC	Surgical phase recognition	(39-41)
M2cai16-instrument	2017	15 videos	LC	Surgical instrument detection	(40)
EndoVis 17' kidney	2017	1500 frames	PN in porcine	Kidney boundary detection	(68)
Boundary detection			-		
Nephrec9	2017	1262 videos	PN	Surgical phase recognition	(70)
EndoVis 17' robotic	2017	8 videos	Porcine procedures	Surgical instrument segmentation; Tissue segmentation	(42)
Instrument segmentation					
ITEC LapGyn4	2018	30682 images	GLS	Surgical action recognition; Anatomical structure recognition; Action on anatomy recognition; Instrument count recognition	(15, 100, 101)
EndoVis 18' sub-challenge	2018	30+ videos	CS	Surgical instrument segmentation; Organ segmentation	(42, 55)
EndoVis 19' surgical	2019	30+ videos	LC	Surgical workflow and skill analysis	(102)
Workflow and Skill analysis					
CholecSeg8k	2020	8080 frames	LC	Anatomical structure segmentation; Surgical instrument	(44, 50, 67, 71, 76, 77,
				segmentation	79, 81, 99, 103–106)
SurgAI	2020	461 images	GLS	Anatomical structure segmentation; Surgical instrument	(10)
				segmentation	
GLENDA	2020	25682 frames	GLS	Endometriosis classification; Endometriosis detection	(11, 13, 18, 47, 93)
M2caiSeg	2020	307 images	Other	Anatomical structure segmentation; Surgical instrument segmentation	(75, 99, 104)
CholecT40	2020	40 videos	LC	Surgical action triplet recognition; Surgical action triplet detection/	(29)
				localization; Surgical instrument presence detection; Surgical	
				instrument detection/localization; Surgical action/verb recognition;	
				Surgical target recognition; Surgical phase recognition	
LapSig300	2020	300 videos	CS	Surgical phase recognition; Surgical action recognition	(86)
Endoscapes dataset	2021	201 videos	Other	Surgical scene segmentation; Object detection; Critical view of	(35–37, 45, 50, 55, 55,
CBVasc181	2021	181 images	IC	Callbladder wall vascularity classification	(16, 17)
AutoLaparo	2021	21 ridaaa	LC	Surgical workflow recognition. Languagene motion prediction	(10, 17)
AutoLaparo	2022	21 videos		Instrument and key anatomy segmentation	(37, 03)
CholecT45	2022	45 videos	LC	Surgical action triplet recognition; Surgical action triplet detection/	(31, 32, 43, 60, 107)
				localization; Surgical instrument presence detection; Surgical	
				instrument detection/localization; Surgical action/verb recognition;	
				Surgical target recognition; Surgical phase recognition	
Dresden surgical Anatomy dataset	2023	13195 images	Proctoco-lectomy	Anatomical structure segmentation	(9, 20, 78, 90, 108)
SurgAI3.8K	2023	3800 images	GLS	Anatomical structure segmentation	(65)
CholecT50	2023	50 videos	LC	Same as CholecT45	(31, 58, 59, 84)
Endo700k	2023	700,000+ images	Other	Label not provided	(43)
Cholec80-CVS	2023	80 videos	LC	CVS recognition	(46)

<sup>a</sup>For the "Procedure" column, "Other" means involving multiple types of surgery. LC, laparoscopic cholecystectomy; GLS, gynecologic laparoscopic surgeries; PN, partial nephrectomy; CS, colorectal surgery.

be used to test the performance of tool presence detection and stage identification methods. In addition, the dataset can be used for a variety of other applications including automated surgical video database indexing, real-time surgical procedure monitoring, and optimizing operating room staffing.

In addition to Cholec80, there are several datasets from surgical flow recognition, instrument segmentation challenges, e.g., M2cai16-workflow challenge (39–41) and EndoVis 17' Robotic Instrument Segmentation sub-challenge (42). Moreover, these datasets were also used in the study by re-labeling.

## 3.1.2 Endo700k

Due to the scarcity of publicly available labeled data and the acquisition and labeling process complexity, Endo700k (43) is aimed at alleviating these problems through large-scale self-supervised pre-training. Endo700k is consisted of nine publicly

available endoscopic datasets merged together, containing over 700,000 unlabeled images involving nine types of minimally invasive surgeries, including prostatectomy, cholecystectomy, and gastrectomy. It enables researchers to fine-tune models on large amounts of unlabeled data.

# 3.2 Specialized datasets

## 3.2.1 CholecSeg8k

The CholecSeg8k dataset (44) is based on the Cholec80 dataset, where 17 video clips were carefully selected from 80 videos of cholecystectomy surgeries, and 8,080 image frames were extracted from them for pixel-level annotation. These images include different stages and scenarios of cholecystectomy surgeries. The



CholecSeg8k dataset is finely labeled, with each image being labeled for 13 critical categories.

## 3.2.2 Endoscapes dataset

The endoscape dataset (45) contains 201 LC videos in which frames are sparsely but regularly labeled with segmentation masks, bounding boxes, and CVS assessments. This dataset can be used for machine learning tasks, including object detection, instance segmentation, CVS prediction, and diverse experiments such as hybrid supervised, semi-supervised, and temporal modeling. The Endoscapes dataset provides four subsets covering different annotation types and data. In addition, comprehensive performance benchmarks are available, providing evaluation criteria for research.

In addition to Endoscapes, Rios et al. (46) annotated all videos in Cholec80 based on the CVS standard, which can also be applied in studies of CVS prediction.

## 3.2.3 GLENDA

GLENDA (Gynecologic Laparoscopic Endometriosis Dataset) is a dataset identifying endometriosis (47). The dataset contains over 350 labeled images of endometriosis lesions covering the four pathologic endometriosis categories and non-pathologic control example images from more than 100 gynecologic laparoscopic procedures. Tasks including binary classification (endometriosis) and detection/localization are supported.

## 3.2.4 CholecT50

CholecT50 (31) is a dataset that identifies action triples for laparoscopic cholecystectomy surgery. The dataset contains 50 video clips documenting laparoscopic cholecystectomy procedures, with 100 categories of surgical action triplets being labeled in detail. These annotations contained approximately 151,000 triad instances, including 6 instrument categories, 10 action verb categories, and 15 target categories.

Cholect45 and Cholect40 are two subsets of Cholect50, containing videos of 45 and 40 surgical procedures, which are also annotated with triad information in the form of [instrument, verb, goal].

## 3.2.5 Dresden surgical anatomy dataset

The Dresden Surgical Anatomy Dataset (DSAD) (9) concentrates on solving the problem of semantic segmentation of abdominal organs.

The dataset provides totally 13,195 laparoscopic images utilizing videos of robot-assisted rectal resections and contains semantic segmentation of eight abdominal organs, the abdominal wall, and two vascular structures. Each image provides a weak annotation of the presence of the organ, providing researchers with various applications of the data.



FIGURE 4

A generalized process for applying DL models to laparoscopic medical scenarios. The first part is the data acquisition and preprocessing steps; the second part is the model architecture and the model evaluation, which is the main discussion in this chapter; the third part is the model output and post-processing operations; and the fourth part is the actual application scenarios.

# 4 Methodology and technical strategy

Most of the studies in the literature collected in this paper have applied DL methods. Figure 4 shows a generic process, containing four main modules: input, model, output and application.

Firstly, videos of laparoscopic surgical need to be collected. In order to process long video data, operations such as frame lifting are usually required. Next, segmentation masks, category labels, bounding boxes, etc. need to be labeled by the surgeon. Subsequently, the data are preprocessed aiming to improve robustness of the model. For different tasks, appropriate network architectures are selected in line with their characteristics, such as Unet and DeeplabV3, and some technical improvement modules, such as the attention mechanism, may be added. It is also possible to adopt different learning paradigms or design entirely new network architectures. Afterwards, the model performance is evaluated. Also, a team of surgeons will perform a qualitative assessment. After the model outputs the results, post-processing operations are usually needed to further enhance the quality of the results. Final result can be applied in different practical scenarios, such as lesion localization and organ identification. These data can also be used for surgical skills training as well as scientific research to advance the field of medicine and artificial intelligence.

As presented in Figure 5, 62% of the included studies are focused on segmentation tasks (73 articles), 25% on classification tasks (32 articles), and 13% on target detection tasks (15 articles). These tasks are detailed. Next, we will analyze and detail the methodological techniques, model architecture, and strategy evaluation for each of these three types of tasks.

# 4.1 Classification methods

This section deals exclusively with publications concentrating on classification methods, where 32 contributions were identified, as presented in Table 2. They will be characterized and logically grouped according to different characteristics in the following. At



Proportion of papers across the three assignment categories and their corresponding evaluation metrics. This figure presents an analysis of research distribution and evaluation metrics in laparoscopic image analysis. (a) Displays the percentage of papers focusing on three types of tasks. (b) Illustrates the frequency of evaluation metrics used for classification tasks. (c) Highlights the frequency of evaluation metrics applied in segmentation tasks. (d) Shows the frequency of evaluation metrics employed for object detection tasks.

first, developments are organized according to learning strategies: supervised and not supervised learning. Then, recent developments in multi-task-based learning are described. Finally, an analysis of evaluation metrics for classification methods.

#### 4.1.1 Supervised learning

Network architectures like the one shown in Figure 6a are usually used in supervised learning methods for classification tasks. The encoder can be a CNN, extracting features with semantic information from the original image through a series of convolutional layers, pooling layers, and other operations. These features retain key information in the image, including organ shape, and texture, and reduce the dimensionality of the data. Classifiers typically have one or more fully connected layers, where the output represents a probability distribution for each category and the one with the highest probability is selected as the classification result.

Practically, ResNet50 is often used as an Encoder network through methods such as transfer learning. Visalaxi et al. (13) and Nifora et al. (11) utilized ResNet50 to classify laparoscopic images to assist in the diagnosis of endometriosis, achieving 90% and over 95% accuracy rates respectively. It is of note that the latter used a larger dataset. For anatomical structure classification, Hayashi et al. (48) introduced a timestamp smoothing technique followed by classification using a Bayesian neural network. Finally, uncertainty-guided temporal filtering based on uncertainty modifies the results with high uncertainty. In addition, Konduri et al. (15) used an improved K-mean algorithm for clustering, and extracted texture and spectral features using local binary patterns and averaged spectra. Feature selection was optimized based on an improved Cuckoo search optimization algorithm. The final classification was performed through a FrCNN, and the results indicated that the method achieved more than 99% accuracy in organ classification.

Furthermore, for CVS prediction, DeepCVS (49) is a two-stage model firstly using Deeplab v3+ with Xception 65 as the backbone of a segmentation network to identify anatomical structures. Then, a multi-label classification network was then utilized to evaluate the achievement of the CVS criterion. Murali et al. (50) proposed to

#### TABLE 2 Classification methods.

References	Year	Learning strategy <sup>a</sup>	Target <sup>c</sup>	Model	Dataset <sup>b</sup>
(15)	2024	SL	Uterus, ovary, liver, colon, etc.	Improved K-means clustering+LBP+AS+Improved cuckoo search optimization+FrCNN	ITEC LapGyn4, Cholec80
(54)	2024	SL	CVS	LG-CVS+Domain Generalization+Disentangled Learning+Latent Graph+GNNs	Endoscapes
(54)	2024	WSL	Stomach, uterus, etc.	ChatGPT 4.0+SceneXplain Plugin	Proprietary
(56)	2023	SL	Triplet	Vision-Language Models+GNNs+Spatio-Temporal Reasoning Network+Visual-Semantic Reasoning Network	CholecT50
(55)	2023	SL	Anatomical structures	SurgicalGPT: GPT2+Language-Vision-GPT+Transformer	Cholec80, EndoVis18
(48)	2023	SL	Areas related to blood vessels	Temporal Label Smoothing+Classification using BNNs +Uncertainty guided temporal filtering	Proprietary
(11)	2023	SL	Endometriosis	ResNet50+Transfer learning	GLENDA
(36)	2023	WSL	CVS	LatentGraph-CVS: Graphical representations containing semantic information about anatomical structures+GNNs	Endoscapes
(37)	2023	USL	CVS	Large-scale self-supervised pretraining+Downstream task fine- tuning: MoCo v2(ResNet50)+TeCNO	AutoLaparo, Cholec80, Endoscapes
(50)	2023	SL	CVS	GNNs+Potential spatio-temporal maps+Per-frame graph prediction+Temporal edge creation	Endoscapes; Cholec80; CholecSeg8k
(59)	2023	SL	Triplet	ResNet50+LSTM+Multi-Label Mutual Channel Loss	CholecT50
(107)	2023	SL	Triplet	Temporal Attention Module+CAGTAM+Encoder-Decoder +ResNet18+CAM	CholecT45
(60)	2023	SL	Triplet	Swin Transfomer+Self-distillation+Multi-task Learning +Ensembling	CholecT45
(61)	2023	SL	Triplet	ResNet50+MCIT-IG(Transformer)+Interaction-Graph(GNNs) +Mixed Supervision	CholecT50
(43)	2023	SL	Triplet	EndoViT(Transformer)+Encoder-Decoder+Transfer learning	Endo700k, Cholec80, CholecT45
(32)	2023	SL	Triplet, CVS, gallbladder inflammation	ConceptNet(GNNs)+Knowledge Graph	CholecT45
(49)	2022	SL	CVS	DeepCVS: Multi-Stage Learning+DeepLabV3 (Xception65) +Multi-Label Classification Networks	Proprietary
(58)	2022	SL	Triplet	ResNet18+CAGAM+MHMA+Encoder-Decoder	CholecT50
(109)	2021	SL	Gallbladder inflammation	ResNet50+Multilevel Bayesian regression models	Proprietary
(110)	2021	SL	CVS	EndoDigest(DNNs)	Proprietary
(13)	2021	SL	Endometriosis	ResNet50+Transfer learning	GLENDA
(17)	2021	USL	GB wall	Multiple-instance learning+Variational Bayesian Gaussian Mixture Models+SVM	GBVasc181
(111)	2020	SL	Unqualifed, pharynx etc.	ResNet50+Inceptionv3+vgg11-bn+vgg16-bn+DenseNet121 +Transfer learning	Proprietary
(16)	2020	SL	GB wall	VGG+ResNet+SVM	GBVasc181
(96)	2020	SL	GB wall	K-Means+SVM+Naïve Bayes+CNNs	Cholec80
(112)	2020	SL	Gallbladder	CNNs	Proprietary
(41)	2020	USL	Abdominal wall, fat tissue etc.	Multi-Instance Multi-Label Learning+Variational Bayesian gaussian mixture models	M2cai16-workflow
(29)	2020	SL	Triplet	MTL+Class Activation Guide+3D Interaction Space	CholecT40
(14)	2018	SL	Uterus, ovaries, liver, colon etc.	AlexNet+GoogLeNet+SVM	Proprietary
(113)	2018	SL	Six porcine tissues: liver, spleen etc.	Hyperpixel classification strategy based on texture and reflectivity information	Proprietary
(114)	2017	SL	Ureteral etc.	GoogLeNet	Proprietary
(115)	2017	SL	Organ tissue	Multi-spectral texture analysis	Proprietary

<sup>a</sup>For the "Learning Strategy" column. SL, supervised learning; WSL, weakly supervised learning.

<sup>b</sup>For the "Dataset" column, "Proprietary" means that the dataset is not publicly available.

<sup>c</sup>Target may involve surgical instruments, etc. Only anatomical structures are listed here.

encode surgical videos as potential spatio-temporal maps representing anatomical structures and instruments and their properties evolving over time. A pre-trained model is first used to predict the graph for each frame. Then, the temporal edges are added between nodes based on spatial consistency and visual and semantic similarity. The article introduces a graph editing module utilizing a priori knowledge and temporal consistency to correct errors in the graphs, which can therefore improve the performance of downstream tasks. The experimental results of this article are superior to DeepCVS (49).

In recent years, the development of Large-scale language modeling(LLM) has been very rapid (51), and Vision-Language Models (VLM) have shown unprecedented potential for understanding complex surgical scenarios (52,



53). Hirides et al. (54) analyzed 100 laparoscopic surgical images directly using ChatGPT4 and its image recognition plugin SceneXplain, and their results performed well for the task of recognizing anatomical structures. SurgicalGPT (55) is an end-to-end trained VLM of gpt for visual questionanswering tasks in surgical scenarios. The model extends the GPT2 model to include visual input and introduces a feature extractor and visual token embedding. By ordering word tokens before visual tokens, the model mimics the way humans think about understanding questions, and thus better infers answers based on images. Experimental results show that SurgicalGPT performs well in anatomical structure classification. Chain-of-Look (56) Prompting is an end-to-end surgical triad recognition method. The method decomposes the task into interpretable steps by constructing a series of video reasoning processes and utilizes a largescale VLM for visual cue generation. In addition, the article introduces a verb-centric modeling scheme to emphasize the core semantic information of surgical actions. The method achieves optimal performance on the CholecT50 dataset.

#### 4.1.2 Unsupervised learning

Unsupervised learning methods are mainly applied to the CVS prediction task. These methods can effectively lower the dependence on a large amount of manually annotated data and contribute to improving the generalization ability and robustness of the model in different surgical scenarios.

Murali et al. (36) used bounding box annotations to train CVS prediction models. During the first stage, key anatomical structures are identified and used as nodes of the graph. Next, the relationships between nodes are predicted to form the graph's edges. In the second stage, GNNs is applied to predict the CVS, and an auxiliary reconstruction goal is introduced to fine-tune the rest of the model. Alapatt et al. (37) achieved end-to-end prediction of CVS. The ResNet-50 feature extractor was firstly pre-trained using Momentum Contrast. This contrast learning method learns image representations through minimizing the embedding differences between different augmented views of the same image and by maximizing the embedding distance between different images. Then initialize the classifier and fine-tuned based on the Endoscapes dataset to predict the CVS.



#### 4.1.3 Multi-task learning

MTL aims to enhance the generalization performance of individual tasks and the overall model by learning multiple related tasks simultaneously. As displayed in Figure 7, in MTL, models are designed to address multiple tasks simultaneously rather than being trained independently for each task separately (57). These tasks usually show some correlation and thus can share the underlying feature representation, which can utilize the limited data resources in a more efficient way. Recent studies have shown that applying multi-task learning to surgical action recognition triad recognition can obviously improve the performance and robustness of the model.

Tripnet (29) is based on the MTL strategy and contains three branches including tool, verb, and target recognition. In this case, the class activation guidance unit is utilizing the weak localization information in the tool prediction, i.e., the class activation map, to guide the recognition of verbs and targets. And a new trainable 3D interaction space is proposed for capturing the associations between the triples components. However, there is still room for improvement in Tripnet's fine-grained performance when dealing with complex scenes. To address this issue, Rendezvous (58) introduces two different attention mechanisms. One of them refers to the Class Activation-Guided Attention Mechanism based on Tool Activation Graphs, capturing spatial attention of the components of a single action triad in a scene for improved verb and target detection. The Multi-Headed Hybrid Attention module is a hybrid of self-attention and cross-attention, capturing the interaction and semantic relationships between tools, verbs, and targets.

To further enhance the extraction of spatio-temporal dependent features, the multi-task fine-grained spatio-temporal framework (MT-FiST) (59) was proposed by Li et al. The model contains four task branches for recognizing surgical tools, actions, targets, and triples. MT-FiST decouples global task features into category-aligned fine-grained features using a multi-labeled intercommunication channel loss function, which can enable the model to learn more local details in the surgical scene and ensure feature differentiation and diversity. And captures the temporal correlation between neighboring frames through a

partially shared parameter LSTM cell. Yamlahi et al. (60) presented the first attempt to use self-distillation to address the problems of category imbalance and label ambiguity in surgical video analysis. At first, an instructor model is trained. Then, a student model is trained using soft labels to reduce the overconfidence problem. Second, a heterogeneous integration model is proposed, which can combine three different Swin Transformer sizes.

Recent studies have introduced GNNs that utilize graph structures to capture the complex relationships between elements during surgery. The MCIT-IG model (61) is consisted of two phases. In the first phase, embeddings for each target category are generated. In the second stage, a bidirectional dynamic graph is constructed to simulate the interaction between instrument instances and target category embeddings, and verbs are learned on the interaction edge features to detect triples. In addition, a hybrid supervised learning strategy that combines weak target presence labeling from MCIT and pseudo-triad labeling from IG is used to train the network. Concept Graph Neural Networks (32) incorporates knowledge graphs into surgical video analysis and models different concepts and their relationships during surgical procedures as graph structures. Elements of the surgical process (e.g., surgical tools, organization, etc.) can be modeled as nodes of the graph. Interactions between the elements are modeled as hyperedges of the graph.

#### 4.1.4 Evaluation metrics

Evaluation metrics are used to evaluate model performance and quantify how well a model performs in different dimensions, helping comprehend the strengths and weaknesses of a model. Totally ten categories of evaluation metrics were used in the classification methodology, each concentrating on a different evaluation dimension.

As shown in Figure 5, Acc and AP were used most frequently, 15 and 11 times, respectively. Acc evaluates the overall correct classification rate of the model, i.e., the number of correctly predicted samples as a proportion of the total number of samples. The high frequency of Acc usage may reflect the fact that overall correctness is still the most intuitive and popular evaluation criterion for a lot of tasks, especially when the class distribution is balanced. However, AP evaluates the balance between the Precision(Pre) and Recall(Rec) of the model under different thresholds, which is particularly appropriate for evaluating the performance of multi-class classification problems. The high-frequency use of AP suggests that researchers value the combined performance of the model under different confidence thresholds, particularly for applications in complex scenarios. The widespread use of Acc and AP may also be associated with their implementation and standardization in mainstream machine learning libraries, such as the support of these metrics in tools like scikit-learn, TensorFlow, and PyTorch.

Moreover, the other metrics were used relatively infrequently. In this study, pre and F1 scores were used 3 times each, and Rec was used 5 times. Besides, acc is simple and easy to understand but performs poorly on class-imbalanced datasets. Pre and Rec are suitable for class-imbalanced scenarios, especially when the cost of focusing on misclassification is high. Pre-high precision suggests fewer false positives but may miss them. Rec high recall indicates fewer misses but may miss them. F1 Score is used for balancing Pre and Rec, and is suitable for scenarios requiring a combination of the two. Krippendorff's alpha (K-Alpha), Sensitivity (Sen), Specificity (Spe), and AUC Score were also used in specific applications, with each metric used once.

## 4.2 Segmentation methods

The current section explains the publications of segmentation methods, including 73 articles, as shown in Table 3. Firstly, they are grouped in line with learning strategies: supervised learning, semi-supervised learning, weakly supervised learning, and unsupervised learning. Subsequently, relevant studies based on transfer learning and attention mechanisms are shared. Finally, evaluation metrics of segmentation methods are analyzed.

#### 4.2.1 Supervised learning

The researchers used various deep-learning models and techniques to investigate the segmentation task. These models cover general-purpose semantic segmentation networks, including the DeepLabv3 family, Mask R-CNN, FPN, and models commonly used for biomedical image segmentation, including Unet and ESFPNet. The selection of models needs to consider their accuracy, speed, complexity, and performance on specific tasks.

The performance of DeepLabV3 in image semantic segmentation tasks has been extensively validated. The network employs techniques such as Dilated Convolution and Global Average Pooling to efficiently extend the receptive field and maintain detailed information, contributing to better capturing organ boundaries and fine texture information. Igaki et al. (62) performed semantic segmentation of sparse connective tissue by DeepLabV3+. AutoLaparo (63) used Mask R-CNN, YOLACT, and YolactEdge to segment anatomical structures, achieving good segmentation results.

When compared with the previous studies, the following articles focus on the application of networks such as Unet that

are commonly used for biomedical image segmentation and are designed to be more specific to the characteristics of medical images and the needs of surgical tasks. Unet (64) is known for its unique encoder-decoder architecture and hopping connection design. A similar network architecture is usually used in segmentation tasks, as shown in Figure 6b. In this case, the encoder employs a representative structure of CNNs, which gradually extracts image features through multiple convolutional and pooling layers to map the input image to a low-resolution feature map. Different from traditional decoders, Unet's decoder employs operations such as up-sampling and convolutional transposition to gradually restore the low-resolution feature map to the original input image's size, helping refine the segmentation results by learning contextual information and detailed features. The jump connection in the Unet architecture connects the feature maps of each layer in the encoder to the corresponding decoder layer, which realizes cross-layer information transfer to better comprehend the semantic information of the image and improve the accuracy and robustness of segmentation.

SurgAI3.8K (65) is the first gynecological dataset with anatomical annotations, on which the authors employed the U-Net architecture to automatically perform the segmentation of the uterus, uterine contours, and regions of the left and right tubal junctions in surgical images. Bardozzo et al. (66) used a U-Net model for semantic segmentation of the liver, which was interpreted a posteriori by Grad-CAM and Grad-CAM++. Additionally, Silva et al. (67), compared the performance of different networks including Unet, Unet++, DynUNet, UNETR, and DeepLabV3+ on the CholecSeg8k dataset. The results demonstrate that the performance of different networks varies on segmentation tasks with different anatomical structures. This indicates that for a specific task, it is essential to consider the advantages and disadvantages of different models and choose the most suitable one.

Numerous studies have used U-Net or its variants as the basic segmentation network architecture, building on it with structural improvements or combining it with other approaches, such as adversarial training strategies, discriminator networks, or the incorporation of new loss functions.

For example, for the application in kidney edge detection, KiBo-Net (68) improves the structure of U-Net by adding additional convolutional and dropout layers. And the input of the network is modified to be the distance field, and by extracting the depth information and distance field of the image, it predicts whether the pixel belongs to the kidney edge or not. Franccois et al. (69) applied the U-Net architecture to detect the occluded contours of the uterus, proposed a new distance-based evaluation score, and enhanced the performance of the network by introducing a new structural penalty term. NephCNN (70) segment blood vessels in laparoscopic nephrectomy videos. The network utilizes a 3D fully convolutional neural network (FCNN) as a segmenter to extract spatio-temporal features and enhance temporal continuity between pixels. An adversarial training strategy is employed to maintain the coherence of the vessel shape by constraining the segmentation results through a discriminator network. The

## TABLE 3 Segmentation methods.

References	Year	Learning strategy <sup>a</sup>	Target <sup>c</sup>	Model	Dataset <sup>b</sup>
(108)	2024	SL	Ureter, spleen,pancreas, etc.	SegFormer+Class Weight Adjustment	DSAD
(104)	2024	WSL	AW, liver, fat ,etc.	Segment Anything (SAM)+SegGPT	m2caiSeg, CholecSeg8k
(99)	2024	SSL	Liver etc.	Autoencoder+SegFormer	m2caiSeg, Cholec80, CholecSeg8k
(116)	2024	SL	Uretersm, nerves, etc.	Unet	Proprietary
(117)	2024	SL	Gallbladder, spleen, etc.	TernausNet++ TernausResNet+LinkNet+DeepLabV3	Proprietary
(105)	2024	SL	Hepatic vein, CBD, etc.	Encoder-Decoder+U-Net5ed+SegNet-VGG19+SegNet-VGG19 +DeepLabv3+	CholecSeg8K
(118)	2024	SL	Uterus, ovaries, fallopian tubes, etc.	Densely Multi-scale Pyramid Module+Feature Fusion Module +Encoder-Decoder	Proprietary
(119)	2024	SL	Contours, bulges and ligaments	SAM+ResNet+DPE+SGA+BFU	Proprietary
(103)	2024	SL+USL	Liver etc.	DeepLab+HRNet32+HRNet48+Swin Transformer Small+Swin Transformer Base	CholecSeg8k
(97)	2024	SL	CVS	DeepCVS	Cholec80
(98)	2024	SL	Gallbladder, CBD, etc.	Channel Attention Pyramid Scene Parsing Plus Network: Pyramid Scene Parsing Plus Module+Multi-scale information Fusion Module Transfer learning	Cholec80
(120)	2024	SL	SMV, ICA, ICV etc.	DL model	Proprietary
(121)	2024	SL	Ureter	UreterNet+FPN	Proprietary
(106)	2024	SL	Liver etc.	SP-TCN+HRNetv2+Swin Transformer	CholecSeg8k
(101)	2024	SL	Endometriosis, CVS, etc.	DeepPyramid++PVF+DPR+VGG16+ResNet34+ResNet50	ITEC LapGyn4
(122)	2024	SL	Gallbladder, intestinal, etc.	Hierarchical Semantic Segmentation (HSS)+Hiera-Mix+Swin Seg	Proprietary
(54)	2024	SL	CVS	LG-CVS+Domain Generalization+Disentangled Learning +Latent Graph+GNNs	Endoscapes
(88)	2024	SL	Hepatic veins, glisson	Feature Pyramid Network(EfficientNetV2-L)+Transfer learning	Proprietary
(123)	2024	SL	Ureteral	Unet	Proprietary
(90)	2024	SL	11 anatomical structures	DeepLabV3, SegFormer+Attention+Multi-Teacher knowledge distillation+Integrated learning	DSAD
(19)	2024	SL	Liver, gallbladder, spleen, etc.	LDCNet: Res2Net+Attention+Encoder-Decoder+FCN	Proprietary
(79)	2024	SSL	Liver etc.	Multi-scale Projection Head	CholecSeg8k
(78)	2024	SSL	11 anatomical structures	DeepLabV3(ResNet50)	DSAD
(24)	2023	SL	16 different target structures	DeepLabv3(ResNet50)+Transfer learning	Proprietary
(124)	2023	SL	Left adrenal vein	ESFPNet+Transfer learning+Encoder-Decoder	Proprietary
(125)	2023	SL	Kidney	A GPU-based pixel-by pixel connectivity segmentation mechanism	Proprietary
(126)	2023	SL	Hepatic vein, glisson	DeepLabV3	Proprietary
(127)	2023	SL	Hepatic vein, glisson	DeepLabV3	Proprietary
(128)	2023	SL	Autonomic nerves	DeepLabV3+(Xception)+Transfer learning	Proprietary
(72)	2023	SL	Go/No-Go zones	GoNoGoNet	Proprietary
(65)	2023	SL	Uterine contour and tubal junction region	Unet	SurgAI3.8K
(20)	2023	SL	AW, colon etc.	DeepLabV3+SegFormer+Transfer learning+Attention	DSAD
(71)	2023	SL	Organs and tissues	LinkNet34+Decoder	CholecSeg8k
(84)	2023	USL		MoCo v2, SimCLR, DINO, SwAV	Endoscapes
(81)	2023	USL	AW, liver, fat, GB	Masked Frequency Consistency module+Segformer, DeepLabV2 +Attention	CholecSeg8k
(40)	2023	SSL	CD, cystic artery	Unet+Multi-scale Projection Head+Auto-Encoder+Hybridloss function	Cholec80, M2cai16-tool, M2cai16-workflow
(62)	2022	SL	Areolar tissue	DeepLabv3+	Proprietary
(42)	2022	SL	Intestine	Unet, Vnet+Discriminator network	EndoVis 18'
(73)	2022	SL	Go/No-Go zones	GoNoGoNet	Proprietary
(76)	2022	SSL	Liver etc.	DeepLabv3+(ResNet)+Active learning	CholecSeg8k
(77)	2022	SSL	Liver, etc.	Unet(ResNet18)+K-Means	CholecSeg8k
(83)	2022	USL	CD, cystic artery	ResNet101+Label relaxation+Pseudo-label+FCN	Proprietary
(86)	2022	SL	mesenteric artery	DeepLabV3(ResNeSt-269)+Transfer learning	LapSig300
(87)	2022	SL	Organs	DeepLabV3+(ResNeSt)+UperNet(SwinTransformer)+Domain randomization	Proprietary
(63)	2022	SL	Uterus etc.	Mask R-CNN+YOLACT+YolactEdge	AutoLaparo

(Continued)

#### TABLE 3 Continued

References	Year	Learning strategy <sup>a</sup>	Target <sup>c</sup>	Model	Dataset <sup>b</sup>
(66)	2022	SL	Liver	Unet	Proprietary
(67)	2022	SL	8 structures: AW, liver, fat etc.	UNet+UNet+++DynUNet+UNETR+DeepLabV3+	CholecSeg8k
(49)	2022	SL	CD etc.	DeepCVS: DeepLabV3+Multistage learning+Classification network	Proprietary
(23)	2022	SL	Loose connective tissue	SwinPA-Net: Swin Transformer+Dense Multiplicative Connection+Local Pyramid Attention	LIVis(Proprietary)
(39)	2022	SL	Go zone, No-Go zone, liver etc.	GoNoGoNet+CholeNet: ResNet50+Pyramid Scene Parsing Network+Multi-scale pyramid pooling	Cholec80; M2cai16-workflow
(21)	2021	SL	Loose connective tissue fibers	Unet+Data Augmentation	Proprietary
(129)	2021	SL	Liver etc.	Mask R-CNN	Proprietary
(130)	2021	SL	GI tract, blood, vessels, uterus etc.	CNNs	Proprietary
(35)	2021	SSL	CVS	TCNN: Multiloss learning+Auto-Encoder	Endoscapes
(82)	2021	USL	CD, cystic artery	ResNet101+Label relaxation+Pseudo-label	Proprietary
(70)	2020	SL	Blood vessels	NephCNN: Adversarial+training+FCNN	Nephrec9
(10)	2020	SL	Uterus, ovaries	Mask R-CNN+Transfer learning	SurgAI(Proprietary)
(102)	2020	SL	Liver, fat etc.	Unet+TernausNet+LinkNet+SegNet+FCN	EndoVis 19'
(69)	2020	SL	Occluding contours of the uterus	Unet+new loss function	Proprietary
(131)	2020	SL	Liver, GO zone, NO-GO zone etc.	CNNs	Proprietary
(75)	2020	SSL	Liver, gallbladder, intestine, artery etc.	CNNs+Encoder-Decoder+USL pretrain	m2caiSeg
(132)	2019	SL	Anatomical structures	Algorithm based on the extraction and matching of image features	Proprietary
(80)	2019	WSL	Liver, gastric etc.	DeepLabv3+FCN	Proprietary
(68)	2019	SL	Kidney	KiBo-Net(Unet)	EndoVis 17' Kidney
(74)	2019	SL+SSL	Liver	Unet variant+Knowledge Distillation	Proprietary
(133)	2019	SL	Gallbladder, CD, bile duct	CNNs+Encoder-Decoder+Depthwise Separable Convolution +Flip-Based Subpixel Reconstruction	Proprietary
(134)	2019	SL	Organs	Xception+Encoder	Proprietary
(135)	2017	SL	Liver	Deep residual networks+FCN+Multi-resolution loss function	Proprietary
(136)	2016	SL	Organs	Superpixel Extraction+Descriptor+Classifier	MICCAI 2014
(137)	2016	SL	Liver, diaphragm, ligament and tissues	Intra-operative 3D Scene Reconstruction+Segementation using structures from Point Cloud+Labelling with laparoscopic scene cues	Proprietary
(138)	2015	SL	Uterus	Thresholding algorithm+SVM	Proprietary
(139)	2015	SL	Uterus	Gaussian Mixture Model	Proprietary
(140)	2014	SL	Uterus etc.	Gaussian Mixture Model	Proprietary

<sup>a</sup>For the "Learning Strategy" column. SL, supervised learning; WSL, weakly supervised learning.

<sup>b</sup>For the "Dataset" column, "Proprietary" means that the dataset is not publicly available.

<sup>c</sup>Target may involve surgical instruments, etc. Only anatomical structures are listed here.

experimental results of NephCNN significantly outperform 2D U-Net and 3D U-Net.

When dealing with small datasets, Monasterio et al. (42) used U-Net and V-Net as segmentation networks, first synthesizing the erroneous segmentation labels and training a discriminator network to detect errors and produce a dense segmentation error map. Subsequently, the segmentation network is co-trained by minimizing the discriminator prediction error with the standard segmentation loss. Uramoto et al. (71) introduced a second-level decoder on top of the base U-Net structure, which adds semantically similar group segmentation of images as a second-level task. The feature maps of the second-level decoder are also fused into the first-level decoder to enrich the latter's feature representation. The approach achieves better performance.

With the development of DL techniques, more researchers have begun to explore alternative encoder-decoder architectures other than Unet. These improved approaches enhance the model's feature extraction and reconstruction capabilities by introducing new modules and techniques, such as feature pyramid networks and multi-scale feature fusion.

GoNoGoNet (39) is applied to recognize anatomical structures, safe areas, and dangerous areas in LC. The network combines ResNet50 and the pyramid scene parsing network. The pyramid pooling module aggregates feature maps from ResNet50 at four different scales. Afterwards, it is mapped to pixel-level classification scores through a convolutional layer. The whole process can be regarded as an encoder-decoder model. Both Khalid et al. (72) and Laplante et al. (73) evaluated the GoNoGoNet model. The results showed that GoNoGoNet accurately identified safe and dangerous zones in the LC.

#### 4.2.2 Semi-supervised learning

The core idea of semi-supervised learning(SSL) is to combine a limited amount of labeled data with a large amount of unlabeled data in order to improve model performance. Next, the application of semi-supervised learning strategies is explored to segmentation tasks.

Fu et al. (74) compared the performance of supervised and semi-supervised learning methods. A combination of supervised and unsupervised loss and an exponential moving average updating strategy for the teacher network is demonstrated through a semi-supervised mean teacher training paradigm. Higher segmentation accuracy and stability are demonstrated compared with the Unet-based supervised network.

However, the issue of insufficient labeled data remains a challenge. To address this, m2caiSeg (75) employed unsupervised pre-training and data augmentation techniques. Despite the good performance on some categories, there is still room for the improvement of the performance for rare categories. To further enhance performance, TCNN (35) utilizes spatially and temporally supervised signals provided by a self-encoder network incorporating temporal cues and anatomical constraints. The framework demonstrates how a low-dimensional representation of the prediction mask can improve performance while maintaining low computational costs.

Recognizing the need for more effective sample selection for labeling, Qiu et al. (76) introduced an active learning method called class-level confidence-aware active learning. The method selects the most informative samples by keeping a class-level confidence bank and combining the confidence scores. The method can achieve better segmentation with a limited labeling budget through effectively utilizing the unlabeled dataset. Similarly, ALGES (77) is also an active learning method selecting the most representative and diverse samples by calculating the predictive segmentation of unlabeled images and the gradient of the model parameters, reducing the labeling workload and improving the model performance.

To maximize the use of existing labels, Jenke et al. (78) trained a surgical scene segmentation model by combining multiple partially annotated datasets. This method incorporated supplemental annotations during model training, significantly improving DC scores and reducing confusion between categories. In addition, Zhang et al. (79) proposed a class-level contrast learning method that introduces a multi-scale projection header and improves the partitioning of the positive sample pairs to learn the contrast of the extracted features at each scale. The model is trained using both segmentation and classification labels. Even though only a relatively small number of labels (1%– 10%) exhibit high intersection-unification (IoU) scores.

#### 4.2.3 Weakly supervised learning

Weakly Supervised Learning (WSL) is also one of the vital methods that can be used to address the challenge of labeling laparoscopic images. In SL, the training data is labeled with exact labels, providing clear guidance for the model. In semisupervised learning (SSL), the model is trained on a small portion of labeled data combined with a larger amount of unlabeled data to leverage both types of information. Meanwhile, in USL, the training data is entirely unlabeled, requiring the model to detect patterns without any label guidance. WSL, however, is characterized by training data with partially accurate or incomplete labels, offering a flexible solution when precise labeling is challenging or costly.

Fuentes et al. (80) proposed a novel method for labeling laparoscopic images, using "stripes" as a weak annotation and combining it with a partial cross-entropy loss function to train a FCNN for scene segmentation. According to experimental results, the segmentation accuracy of the method is close to that of a fully supervised method on three different datasets, while the time required for labeling is reduced by approximately 13 times.

#### 4.2.4 Unsupervised learning

Unsupervised learning (USL) is an approach in machine learning which is opposed to supervised learning. In unsupervised learning, the training data does not contain labels or results. Meanwhile, the algorithm needs to figure out the hidden patterns and structures in the data itself. Next, we will delve into how unsupervised learning can be used for segmentation tasks.

The Masked Frequency Consistency (MFC) module (81) is employed to solve the problem of domain adaptive semantic segmentation of laparoscopic images. The module is implemented by image frequency representation, masking strategy and consistency regularization. The MFC method is demonstrated to be comparable to fully supervised methods without manual annotation, facilitating knowledge transfer from computer simulations to real laparoscopic datasets as well as enabling model generalization across domains.

Next, the following study explores self-supervised learning, an approach that utilizes the structure and intrinsic relationships of the data itself for feature learning. Owen et al. (82, 83) applied both label relaxation and pseudo-label self-supervision strategies. The label relaxation method will transform the traditional segmentation problem into a heat map regression problem, where the true label heat map is obtained based on the Euclidean distance transform of the original annotation. Moreover, and this method is capable of better dealing with the fuzzy labels of the structure in the image. At first, the pseudo-labeled self-supervised curation method trains a new model in the teacher-student architecture by training an initial model on labeled data and then using the predictions of that model as pseudo-labels for unlabeled data.

Among them, the article Detection (82) is based on FCN and uses ResNet101 as the backbone network. Besides, cross-entropy loss and soft cross-entropy loss are employed to train the model. The method achieved high accuracy and was recognized by surgeons. Instead, Alkhamaiseh et al. (40) combined unsupervised pre-training and supervised fine-tuning. In this, the autoencoder extracts features from a partially prepared dataset and uses these features as pre-training weights for the U-Net encoder layer. Finally, Ramesh et al. (84) evaluated the performance of four state-of-the-art SSL methods including MoCo v2, SimCLR, DINO, and SwAV on a surgical video dataset and also investigated the SSL methods regarding different hyper-parameter settings, data availability, and generalization capabilities, exhibiting their potential in dealing with small datasets and data-scarce domains.

#### 4.2.5 Transfer learning

Transfer learning is a technique widely used in DL, as shown in Figure 8, which can migrate the learned knowledge to a specific task by using models pre-trained on large datasets (e.g., ImageNet). This approach significantly reduces the training time and improves the model's performance in tasks with limited data and performs particularly well in medical image analysis (85). Laparoscopic surgical images suffer from problems including limited data volume and labeling difficulties. This makes it challenging to train models from scratch. In this case, transfer learning provides an efficient solution. That is, the pre-trained model is fine-tuned on surgical images to adapt it to a specific application scenario.

DeepLabV3 is fully pre-trained on large-scale datasets such as ImageNet. Meanwhile, DeepLabV3 features a flexible architecture that can be built based on different backbone networks (e.g., ResNet, ResNeSt, Xception, and EfficientNet.). Therefore, researchers can choose a suitable model according to the demands of specific tasks and the limitations of computational resources. DeepLabV3 and its variants are implemented in numerous opensource frameworks, such as TensorFlow, and PyTorch, providing rich tools and documentation. On this basis, DeepLabV3 becomes a widely used pre-training model for segmentation tasks.

Kitaguchi et al. (86) used DeepLabv3+ as a backbone for migration learning and training on LapSig300, showing that the

model was able to accurately recognize IMAs at over 12 FPS with an mDC of 0.798. The feasibility for real-time navigation of blood vessels in laparoscopic colorectal surgery was demonstrated. Next, Yoon et al. (87) proposed semantic image synthesis using a virtual surgical environment to enhance the performance of surgical scene segmentation. Through the combination of manual synthetic data, domain randomized synthetic data with real data, which provides a wider dataset. The results show that synthetic data can significantly enhance the performance of the models on lowperformance categories.

In addition to deeplabv3, networks such as Mask R-CNN can be used as pre-trained models to provide powerful feature extraction for segmentation tasks.

Madad et al. (10) achieved accurate localization and segmentation of key structures such as the uterus, ovaries, and surgical tools by adopting Mask R-CNN for migration learning. Une et al. (88) developed models for identifying the hepatic veins and glissonean based on a feature pyramid network (FPN) with EfficientNetV2-L as the backbone. The results showed high accuracy and sufficient processing speed.

## 4.2.6 Attention methods

Attention Mechanism enhances the network's ability to focus on specific parts during information processing, similar to the human visual process: when processing a large amount of information, more attention is paid to what is relevant to the task, thus enhancing the efficiency of information utilization. In neural networks, the attention mechanism is implemented by assigning different weights to input features, i.e., determining the region that the model should focus on based on the correlation between input



elements (89). As shown in Figure 9, a commonly used implementation is the dot-product attention mechanism, which measures the relevance of elements by calculating the dot product between query vectors and key vectors, and then generates the attention weights to weight the value vectors so as to focus on key regions. In addition, to enhance the model's representation of different features, the Multi-Head Attention mechanism applies multiple attention heads in parallel and computes the attention independently in different projection spaces, enabling the model to capture information from multiple perspectives. This mechanism has great adaptability and flexibility in dealing with complex data, and is especially suitable for dealing with laparoscopic surgical images with complex and unevenly distributed structures, which can still effectively focus on the key parts when the anatomical structures are deformed and distorted, thus improving the accuracy of the network.

SwinPA-Net (23) used the Swin Transformer as an encoder. Second, efficient information transfer between feature hierarchies can be achieved through dense multiplicative connections. In the meanwhile, the local pyramid attention module helps the network better focus on key regions and aggregate multi-scale features. On the decoder side, a gradual upsampling strategy is applied to maintain the fineness of the segmentation results. SwinPA-Net achieves state-of-the-art performance in three medical image segmentation tasks. Kolbinger et al. (20) trained independent and joint models based on DeepLabv3 and SegFormer for 11 anatomical structures in laparoscopic images. SegFormer's self-attention mechanism can model the dependencies between different locations. Experimental results indicate that the models based on the attention mechanism have higher segmentation accuracy and generalization ability compared to those using only CNNs.

In practical applications, real-time is also an important consideration. To this end, Maack et al. (90) proposed an approach based on multi-teacher knowledge distillation. The mDC score of the real-time network is improved by training multiple DeepLabv3 and SegFormer-based teacher networks and aligning the outputs of these teacher networks with the output of a student network. Lightweight dynamic convolution network (LDCNet) (19) is a lightweight novel dynamic convolution network for real-time segmentation. A dynamic convolutionbased attention module is used between the coders and decoders, allowing the network to better focus on regions of interest. The encoder part uses Res2Net and introduces a sensory wild block module to further refine the features. The progressive decoder obtains effective feature reconstruction by fusing high-level features with low-level features. The experimental results suggest that LDCNet combines high speed and accuracy and exhibits high practical value in real scenarios.

## 4.2.7 Evaluation metrics

Thirteen different evaluation metrics were used in the segmentation task, with significant differences in their frequency of use.

As presented in Figure 5c, the Dice Coefficient (DC) is one of the most commonly used segmentation metrics to measure the similarity between predicted segmentation and true segmentation. DC was used for totally 27 times in the literature, suggesting that researchers place great importance on the quality of the overlap



of segmented regions. IoU is another widely used metric, which was used a total of 24 times, aiming to assess the intersection and concurrency ratio between predicted segmentation and true segmentation.

Acc, F1 Score, Rec, and Pre also have some applications in the segmentation task. Rec measures the model's ability to detect all positive sample regions in the segmentation task, focusing on regions that are not correctly segmented. Pre evaluates the proportion of positive sample regions predicted by the model that are actually positive, emphasizing the accuracy of the prediction. Acc is of comparatively low importance because the segmentation task is more concerned with regions of overlap and matching.

Other metrics, including Sen, correlation, agreement are used less frequently. AP is used in some complex multi-class segmentation tasks. Hausdorff\_95 evaluates the maximum distance between the predicted segmentation and the true segmentation boundary, and is mainly applied in segmentation tasks requiring high accuracy. Mean Misrecognition measures the missegmentation region of the model. Error evaluates the overall segmentation error of the model.

## 4.3 Object detection methods

TABLE 4 Object detection methods

In the current section, we will discuss the research associated with the target detection task, which consists of 15 articles, as shown in Table 4. These studies can be categorized into two main groups: one is based on traditional image processing algorithms, and the other is based on DL methods. Next, we will elaborate on these two directions, first presenting the research

progress	of	traditional	image	processing	algorithms	and	then	
discussin	g th	e latest resu	ılts base	ed on DL.				

# 4.3.1 Based on conventional image processing techniques

The next presentation is a study based on traditional image processing techniques adopted for addressing tasks, including lesion detection. The core idea of traditional image processing techniques is to obtain the localization and identification of the target of interest through mathematical manipulation and feature analysis, mainly including feature extraction and pattern recognition.

Prokopetc et al. (91) trained uterus detectors and FU connection detectors. These detectors incorporate connection-specific context-sensitive features to achieve automatic target detection through linear classification. Nakasuji et al. (92) successfully identified triangular tissue regions of pulled by surgical forceps, by combining corner point detection and ridge detection with Delaunay triangular dissection. Visalaxi et al. (93) extracted focal regions of endometriosis using OpenCV, adaptive thresholding, and contour masking, and evaluated the recognition effect by the mean intensity value.

#### 4.3.2 Based on deep learning

The aforementioned methods based on traditional image processing techniques have been successful to a certain extent, while they have many limitations, such as the comparative sensitivity to image quality and parameter selection, as well as the lack of precision in localizing specific targets. Therefore, the subsequent research incorporated novel techniques, aiming to achieve higher performance.

References	Year	Learning strategy <sup>a</sup>	Target <sup>c</sup>	Method	Dataset <sup>b</sup>
(141)	2024	SL	Liver base of segment IV, Rouviere's sulcus	YOLOv7	Proprietary
(142)	2024	SL	Vas deferens and subabdominal vessels, etc.	YOLOv8	Proprietary
(36)	2023	WSL	Gallbladder, CD, cystic artery, hepatic duct, cystic plate, etc.	LatentGraph-CVS: constructing graphical representations containing semantic information about anatomical structures and visual features +GNNs	Endoscapes
(26)	2023	SL	LM-RS, LM-S4, LM-CBD, LM-CD	YoloV3	Proprietary
(94)	2023	SL	CBD, CD, S4, RS	YoloV3	Proprietary
(95)	2023	SL	EHBD, CD, S4, RS	YoloV3	Proprietary
(25)	2023	SL	Uterus	Comparing the performance of Faster R-CNN, SSD, CenterNet, EfficientDet, YOLOv4, YOLOv5, YOLOv7+transfer learning	Proprietary
(18)	2022	SL	Endometriosis	Faster R-CNN(ResNet50), Mask R-CNN(ResNet101)	GLENDA
(143)	2022	SL	CD, cystic artery, CBD, cystic plate	SurgSmart: YoloV3(ResNet)+transfer learning	Proprietary
(144)	2021	SL	CD, CBD, S4, RS	YoloV3	Proprietary
(145)	2021	SL	Anatomical landmarks	Yolo,YoloV4 tiny	Proprietary
(27)	2021	SL	Vas deferens	YoloV4	Proprietary
(93)	2021	SL	Endometriosis	Conventional image processing techniques	GLENDA
(92)	2017	SL	Triangular tissue	Conventional processing techniques	Proprietary
(91)	2015	SL	FU-junctions	Conventional image processing techniques	Proprietary

<sup>a</sup>For the "Learning Strategy" column. SL, supervised learning; WSL, weakly supervised learning. <sup>b</sup>For the "Dataset" column, "Proprietary" means that the dataset is not publicly available.

<sup>c</sup>Target may involve surgical instruments, etc. Only anatomical structures are listed here.

The DL models in these articles cover a wide range of classical target detection algorithms, such as YOLOv3, and Faster R-CNN. Mask R-CNN employs a two-step detection process. First, a pre-trained ResNet backbone network is used to pull out features. Afterwards, the features are sent through a feature pyramid network for fusion. Then, more Region Proposal Networks (RPNs) propose bounding boxes. RoI Pooling and RoI Align pull out features and perform target detection or pixel-level segmentation. A similar network architecture is usually used in target detection tasks, as shown in Figure 6c. By contrast, the YOLO family of models excels in real-time applications with its high inference speed and good detection performance for scenarios requiring fast processing.

Nakanuma et al. (26) developed a YOLOv3-based AI system for the detection of anatomical marker points that surgeons rely on during surgery. The system's performance was evaluated by an external evaluation committee, showing that the system could accurately identify key anatomical landmarks. Similarly, Fujinaga et al. (94) and Endo et al. (95) conducted a similar study. The results demonstrated the method's effectiveness. In addition to the direct application of YOLOv3, boonkong et al. (25) compared the performance of eight models on uterine detection tasks, including Faster R-CNN, SSD, CenterNet, EfficientDet, YOLOv4, YOLOv5, and YOLOv7. Based on the obtained results, YOLO series models perform optimally in terms of accuracy and speed.

#### 4.3.3 Evaluation metrics

As shown in Figure 5, target detection methods use a wide variety of evaluation metrics, including both qualitative and quantitative evaluations.

Quantitative evaluation is based on numerical computation and image analysis and is mainly adopted for measuring the performance of algorithms in terms of localization and accuracy. Among them, AP is used most frequently. There are some qualitative evaluation methods, such as five-level scale evaluation and questionnaire format evaluation. Moreover, the effectiveness of the algorithm is evaluated from the point of view of user experience and practical application by collecting users' subjective feelings and opinions on the effectiveness of the algorithm.

# **5** Discussion

## 5.1 Datasets

Totally 24 publicly available datasets were involved in the study of anatomical structures in laparoscopic surgical images, among which 7 were generalized datasets and 11 were proprietary datasets. The most frequently used dataset was the generic dataset, which was used for 13 times. Although they were used more frequently, their additional annotation work remained a challenge for research efficiency and resource allocation.

Aside from the high-frequent use of the generic datasets, there was a large variation in the frequency of application of the other datasets, with 15 datasets being used only once. This situation reveals the dispersion and diversity of dataset use within the field, possibly caused by the diversity of research topics and the unique strengths of specific datasets. This diversity allows researchers to select the most appropriate datasets for different problems, driving innovation in methods and techniques. However, this decentralization also brings certain disadvantages that may result in duplication of effort: researchers need to develop or adapt specific processing methods for each new dataset, which can thus increase research time and costs. Secondly, the lack of extensive validation and application of datasets used at low frequencies may hide some undiscovered flaws or limitations, lowering the reliability of the research results.

The comparison of similar tasks reveals that they all employ different datasets. This makes it difficult to directly compare the methods and results of different studies, limits the unified assessment and standardization of techniques in the field, and increases the complexity of comprehensively assessing the effects of different methods.

It is of note that most of the datasets contain annotations not only for anatomical structures but also for tools, surgical stages, and maneuvers. The multiple-annotated datasets provide rich contextual information for research and help develop more comprehensive and intelligent algorithms to enhance the adaptability and accuracy of models in real surgical settings. Additionally, seven datasets were generated from competitions or other events organized by MICCAI. These competition datasets drive the development of technology and innovation in the field through providing a standardized evaluation platform.

Finally, it is vital to emphasize that more than 60 articles used private datasets. Their experimental results are difficult to compare with other methods. More importantly, because private datasets are difficult to access, this can limit the possibility for other research groups to carry out further studies on the data, also resulting in unrepeatable and unvalidated experimental results.

To counter the existing problems, more collaboration and sharing of resources may be needed to improve the overall efficiency of research and the credibility of results. Therefore, we call on future researchers to actively use publicly accessible and comparable datasets to develop and validate their methods, as well as to openly collect and self-label their data so that other studies can access and use these datasets.

## 5.2 Methodology

Regarding methodology, each category of tasks is characterized by its own research methods and trends.

Classification tasks are mainly categorized into supervised learning and unsupervised learning methods. Supervised learning methods dominate the classification task, usually by fine-tuning the DL model or utilizing data enhancement techniques to improve the model performance. Weakly supervised learning and unsupervised learning methods mainly focus on the CVS prediction task. Multitask learning methods are widely applied in surgical action triplet recognition tasks. These methods perform well when handling complex tasks, while for some simpler classification tasks, simple transfer learning has been capable of achieving good results. The vast majority of studies in segmentation tasks depend on manually labeled data, and the use of supervised learning methods is the most common strategy. In supervised learning, transfer learning is widely applied. Meanwhile, attention mechanisms have also become a hot research direction, exhibiting the potential to improve segmentation accuracy. However, due to the high cost of manually labeling data, semi-supervised learning, weakly supervised learning, and unsupervised learning methods are also gradually gaining attention and application as a trend for future development. Graph neural networks are beginning to show promising applications in segmentation tasks, providing novel solutions.

There is relatively little literature related to the target detection task, and the early research is mainly based on traditional image processing techniques, while the research in the last five years is mainly based on DL methods. Most methods directly apply the existing model or are fine-tuned by transfer learning. However, the field of target detection has not yet been significantly developed, and there remains more room for innovation and improvement.

The development of these methods and techniques has brought significant progress and wide application prospects in the field. Nevertheless, there are also a lot of challenges and opportunities. Supervised learning methods, despite their superior performance, rely on a large amount of manually labeled data, and data acquisition and labeling are costly. Although research and application of unsupervised and weakly supervised learning methods can alleviate this problem to a certain extent, their accuracy and stability still need to be addressed. Real-time is another vital challenge, especially in high-risk environments such as surgery, where the inference speed of algorithms directly influences clinical decisions and patient safety. However, only a small portion of the literature focuses on inference speed, and future research needs to focus on the real-time optimization of algorithms to satisfy the needs of clinical applications. And with the continuous development of new technologies, including MTL, attention mechanisms, and GNNs, the performance and application scope of DL methods will be further improved. Moreover, this provides researchers with a wealth of research topics and innovation space, bringing new opportunities for progress in the field.

In addition, large models have achieved introduced attention in various domains, however, in this particular domain, they are limited to a few applications in categorization tasks, but all of them perform well. This demonstrates the great potential and necessity of exploring VLM. The multimodal fusion capabilities of VLM are leveraged in order to enhance the understanding of complex anatomical structures in laparoscopic surgical images. In addition, their strong generalization capabilities and ability to handle data scarcity make them ideal for dealing with high annotation costs and restricted data volumes.

## 5.3 Evaluation metrics

Evaluation metrics exert a vital role, not only in assessing algorithm performance, but also in directly influencing the application and diffusion of algorithms in clinical practice.

Current statistics find that classification, segmentation, and target detection tasks all involve multiple types of evaluation

metrics. There are fewer cases where the tasks are the same and the evaluation metrics are also the same, making it difficult to comprehensively assess and compare the performance of different algorithms. Secondly, in the target detection domain, the number of evaluation metrics is comparable to that of the segmentation task, even though the number of literatures involved is relatively small. This maysuggest that the target detection domain has not been well addressed for the harmonization of evaluation metrics. In summary, we call for the use of more consistent and comprehensive evaluation metrics, aiming to more intuitively assess and compare the performance of different algorithms and to promote further development in the field.

In addition, we note that the target detection task involves qualitative evaluation. This suggests that in addition to quantitative evaluation metrics, it is increasingly vital to consider the actual usage and experience of physicians. Therefore, we suggest including more qualitative evaluation metrics in order to comprehensively assess the applicability and usefulness of the algorithms in a clinical setting. As a result, when selecting evaluation metrics, their correlation with clinical outcomes should be considered to ensure that the model can contribute in practical clinical applications.

# 5.4 Summary of challenges and potential future work

According to the literature we have collected, research in this area has shown a trend of rapid growth from year to year, showing widespread interest and sustained investment of resources.

In terms of datasets, the large number and diversity of types lead to problems of fragmentation and duplication in the use of datasets. While this diversity drives innovation in techniques and methods, it can also increase research time and costs, making it difficult to directly compare the results of different studies. In addition, numerous studies use private datasets, making it difficult to reproduce and validate results. Future studies should encourage researchers to use publicly available and comparable datasets, or to make publicly available self-collected and labeled data, hoping to increase the transparency and reproducibility of studies.

In terms of methodology, different tasks are characterized by different research methods and trends. Classification tasks mainly depend on supervised learning, but weakly supervised learning and unsupervised learning are also gaining attention, especially in CVS prediction tasks. Supervised learning methods are most commonly used in segmentation tasks, but semi-supervised, weakly supervised and unsupervised learning methods are also trending due to the high cost of data labeling. There is less literature on target detection tasks, and not yet significantly developed. Moreover, the application of VLM is relatively small, and future research should attempt to combine it with image analysis of laparoscopic surgery, which is expected to significantly improve the technical level and practical application in this field.

In terms of evaluation metrics, there is a wide variety of evaluation metrics involved in classification, segmentation and target detection tasks. In addition, future research should advocate the use of more consistent and comprehensive evaluation metrics to more intuitively assess and compare the performance of different algorithms. Meanwhile, qualitative evaluation metrics should be added to take into account the actual use and experience of physicians and to establish a correlation with clinical outcomes to ensure the effectiveness of the model in practical applications.

The continued growth and technological innovations in this field of research have exerted a profound impact on clinical practice. As a growing body of studies focus on solving realworld surgical challenges, we can anticipate the emergence of more accurate and smarter surgical assistance systems and tools in future clinical practice. These systems and tools will significantly improve the precision and safety of surgery, reduce complications and surgical risks, and therefore, provide better outcomes and quality of life for patients.

# 6 Conclusion

This study provided an overview of recent developments in the field of classification, segmentation and target detection of anatomical structures in laparoscopic images. The core subtasks and their applications in real medical scenarios were first discussed, followed by a statistical analysis of the current state of use of the dataset. Then, the methods, models, and evaluation metrics used in the literature were thoroughly analyzed and discussed, offering insights and reflections on current research. Finally, directions and strategies for future development were proposed to address the existing shortcomings and challenges, hoping to foster further development and innovation in the field.

# Author contributions

RZ: Investigation, Writing – original draft. DW: Methodology, Writing – original draft. HZ: Conceptualization, Writing – review & editing. YZ: Methodology, Writing – original draft. LZ: Conceptualization, Supervision, Writing – review & editing. TC: Investigation, Writing – review & editing. WL: Resources,

## References

1. Buia A, Stockhausen F, Hanisch E. Laparoscopic surgery: a qualified systematic review. World J Methodol. (2015) 5(4):238. doi: 10.5662/wjm.v5.i4.238

2. Varela JE, Wilson SE, Nguyen NT. Laparoscopic surgery significantly reduces surgical-site infections compared with open surgery. *Surg Endosc.* (2010) 24:270–6. doi: 10.1007/s00464-009-0569-1

3. Rao PP, Rao PP, Bhagwat S. Single-incision laparoscopic surgery-current status and controversies. J Minim Access Surg. (2011) 7(1):6–16. doi: 10.4103/0972-9941. 72360

4. Bernhardt S, Nicolau SA, Soler L, Doignon C. The status of augmented reality in laparoscopic surgery as of 2016. *Med Image Anal.* (2017) 37:66–90. doi: 10.1016/j. media.2017.01.007

5. Ward TM, Mascagni P, Ban Y, Rosman G, Padoy N, Meireles O, et al. Computer vision in surgery. Surgery. (2021) 169(5):1253-6. doi: 10.1016/j.surg.2020.10.039

6. Guo K, Tao H, Zhu Y, Li B, Fang C, Qian Y, et al. Current applications of artificial intelligence-based computer vision in laparoscopic surgery. *Laparosc Endosc Rob Surg.* (2023) 6(3):91–6. doi: 10.1016/j.lers.2023.07.001

Writing – review & editing. ZY: Project administration, Writing – review & editing.

# Funding

The authors declare that no financial support was received for the research and/or publication of this article.

# Acknowledgments

Thanks to Servier for the image "drug-capsule-4," which is released under the CC-BY 3.0 Unported license and is source material for Figure 2. It can be viewed at https://bioicons.com/. In accordance with the license requirements, it is hereby declared that Servier assumes no responsibility for the contents and conclusions of this paper.

# Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

## **Generative AI statement**

The author(s) declare that no Generative AI was used in the creation of this manuscript.

# Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

7. Kitaguchi D, Takeshita N, Hasegawa H, Ito M. Artificial intelligence-based computer vision in surgery: recent advances and future perspectives. *Ann Gastroenterol Surg.* (2022) 6(1):29–36. doi: 10.1002/ags3.12513

8. Choi B, Jo K, Choi S, Choi J. Surgical-tools detection based on convolutional neural network in laparoscopic robot-assisted surgery. In: 2017 39th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC). IEEE (2017). p. 1756–9.

9. Carstens M, Rinner FM, Bodenstedt S, Jenke AC, Weitz J, Distler M, et al. The dresden surgical anatomy dataset for abdominal organ segmentation in surgical data science. *Sci Data*. (2023) 10(1):1–8. doi: 10.1038/s41597-022-01719-2

10. Zadeh SM, Francois T, Calvet L, Chauvet P, Canis M, Bartoli A, et al. SurgAI: deep learning for computerized laparoscopic image understanding in gynaecology. *Surg Endosc.* (2020) 34:5377–83. doi: 10.1007/s00464-019-07330-8

11. Nifora C, Chasapi L, Chasapi M-K, Koutsojannis C. Deep learning improves accuracy of laparoscopic imaging classification for endometriosis diagnosis. (2023).

12. Duffy JM, Arambage K, Correa FJ, Olive D, Farquhar C, Garry R, et al. Laparoscopic surgery for endometriosis. *Cochrane Database Syst Rev.* (2020) 10(10): CD011031. doi: 10.1002/14651858.CD011031.pub2

13. Visalaxi S, Sudalai Muthu T. Automated prediction of endometriosis using deep learning. Int J Nonlinear Anal Appl. (2021) 12(2):2403–16. doi: 10.22075/ijnaa.2021.5383

14. Petscharnig S, Schöffmann K. Learning laparoscopic video shot classification for gynecological surgery. *Multimed Tools Appl.* (2018) 77:8061–79. doi: 10.1007/s11042-017-4699-5

15. Konduri PSR, Rao GSN. Full resolution convolutional neural network based organ and surgical instrument classification on laparoscopic image data. *Biomed Signal Process Control.* (2024) 87:105533. doi: 10.1016/j.bspc.2023.105533

16. Loukas C, Frountzas M, Schizas D. Patch-based classification of gallbladder wall vascularity from laparoscopic images using deep learning. *Int J Comput Assist Radiol Surg.* (2021) 16:103–13. doi: 10.1007/s11548-020-02285-x

17. Loukas C, Gazis A, Schizas D. A multiple-instance learning approach for the assessment of gallbladder vascularity from laparoscopic images. *arXiv* [Preprint]. *arXiv:2107.12093* (2021).

18. Leibetseder A, Schoeffmann K, Keckstein J, Keckstein S. Endometriosis detection and localization in laparoscopic gynecology. *Multimed Tools Appl.* (2022) 81(5):6191–215. doi: 10.1007/s11042-021-11730-1

19. Yin Y, Luo S, Zhou J, Kang L, Yu-Chian Chen C. Ldcnet: Lightweight dynamic convolution network for laparoscopic procedures image segmentation. *Neural Netw.* (2024) 170:441–52. doi: 10.1016/j.neunet.2023.11.055

20. Kolbinger FR, Rinner FM, Jenke AC, Carstens M, Krell S, Leger S, et al. Anatomy segmentation in laparoscopic surgery: comparison of machine learning and human expertise–an experimental study. *Int J Surg.* (2023) 109(10):2962–74. doi: 10.1097/JS9.00000000000595

21. Kumazu Y, Kobayashi N, Kitamura N, Rayan E, Neculoiu P, Misumi T, et al. Automated segmentation by deep learning of loose connective tissue fibers to define safe dissection planes in robot-assisted gastrectomy. *Sci Rep.* (2021) 11:1–10. doi: 10.1038/s41598-021-00557-3

22. Di Buono G, Buscemi S, Cocorullo G, Sorce V, Amato G, Bonventre G, et al. Feasibility and safety of laparoscopic complete mesocolic excision (CME) for right-sided colon cancer: short-term outcomes. A randomized clinical study. (2021).

23. Du H, Wang J, Liu M, Wang Y, Meijering E. Swinpa-net: Swin transformerbased multiscale feature pyramid aggregation network for medical image segmentation. *IEEE Trans Neural Netw Learn Syst.* (2022) 35(4):5355–66. doi: 10. 1109/TNNLS.2022.3204090

24. Kolbinger FR, Bodenstedt S, Carstens M, Leger S, Krell S, Rinner FM, et al. Artificial intelligence for context-aware surgical guidance in complex robot-assisted oncological procedures: an exploratory feasibility study. *Eur J Surg Oncol.* (2023) 50(12):106996. doi: 10.1016/j.ejso.2023.106996

25. Boonkong A, Khampitak K, Hormdee D. Performing and evaluation of deep learning models for uterus detection on soft-tissue cadavers in laparoscopic gynecology. *IEEE Access.* (2023) 11:72027–36. doi: 10.1109/ACCESS.2023.3293006

26. Nakanuma H, Endo Y, Fujinaga A, Kawamura M, Kawasaki T, Masuda T, et al. An intraoperative artificial intelligence system identifying anatomical landmarks for laparoscopic cholecystectomy: a prospective clinical feasibility trial (j-summit-c-01). *Surg Endosc.* (2023) 37(3):1933–42. doi: 10.1007/s00464-022-09678-w

27. Cui P, Zhao S, Chen W. Identification of the vas deferens in laparoscopic inguinal hernia repair surgery using the convolutional neural network. *J Healthc Eng.* (2021) 2021(1):5578089. doi: 10.1155/2021/5578089

28. Katić D, Wekerle A-L, Gärtner F, Kenngott H, Peter Müller-Stich B, Dillmann R, et al. Knowledge-driven formalization of laparoscopic surgeries for rule-based intraoperative context-aware assistance. In: *Information Processing in Computer-Assisted Interventions: 5th International Conference, IPCAI 2014, Fukuoka, Japan, June 28, 2014. Proceedings 5.* Springer (2014). p. 158–67.

29. Innocent Nwoye C, Gonzalez C, Yu T, Mascagni P, Mutter D, Marescaux J, et al. Recognition of instrument-tissue interactions in endoscopic videos via action triplets. In: Medical Image Computing and Computer Assisted Intervention–MICCAI 2020: 23rd International Conference, Lima, Peru, October 4–8, 2020, Proceedings, Part III 23. Springer (2020). p. 364–74.

30. Innocent Nwoye C, Alapatt D, Yu T, Vardazaryan A, Xia F, Zhao Z, et al. CholecTriplet2021: a benchmark challenge for surgical action triplet recognition. *Med Image Anal.* (2023) 86:102803. doi: 10.1016/j.media.2023.102803

31. Nwoye CI, Yu T, Sharma S, Murali A, Alapatt D, Vardazaryan A, et al. Cholectriplet2022: show me a tool and tell me the triplet—an endoscopic vision challenge for surgical action triplet detection. *Med Image Anal.* (2023) 89:102888. doi: 10.1016/j.media.2023.102888

32. Ban Y, Eckhoff JA, Ward TM, Hashimoto DA, Meireles OR, Rus D, et al. Concept graph neural networks for surgical video understanding. *IEEE Trans Med Imaging*, (2023) 43(1):264–74. doi: 10.1109/TMI.2023.3299518

33. Strasberg SM, Brunt ML. Rationale and use of the critical view of safety in laparoscopic cholecystectomy. *J Am Coll Surg.* (2010) 211(1):132–8. doi: 10.1016/j. jamcollsurg.2010.02.053

34. Ahmed Alturfi RR, Hilmi AA. Cystic duct identification in laparoscopic cholecystectomy review articles. *Saudi J Laparosc.* (2020) 5(1):6–11. doi: 10.4103/SJL.SJL\_5\_20

35. Alapatt D, Mascagni P, Vardazaryan A, Garcia A, Okamoto N, Mutter D, et al. Temporally constrained neural networks (TCNN): A framework for semi-supervised video semantic segmentation. *arXiv* [Preprint]. *arXiv:2112.13815* (2021).

36. Murali A, Alapatt D, Mascagni P, Vardazaryan A, Garcia A, Okamoto N, et al. Latent graph representations for critical view of safety assessment. *IEEE Trans Med Imaging.* (2023) 43(3):1247–58. doi: 10.1109/tmi.2023.3333034

37. Alapatt D, Murali A, Srivastav V, Mascagni P, Consortium A, Padoy N. Jumpstarting surgical computer vision. *arXiv* [Preprint]. *arXiv:2312.05968* (2023).

38. Twinanda AP, Shehata S, Mutter D, Marescaux J, De Mathelin M, Padoy N. Endonet: a deep architecture for recognition tasks on laparoscopic videos. *IEEE Trans Med Imaging*. (2016) 36(1):86–97. doi: 10.1109/TMI.2016.2593957

39. Madani A, Namazi B, Altieri MS, Hashimoto DA, Maria Rivera A, Pucher PH, et al. Artificial intelligence for intraoperative guidance: using semantic segmentation to identify surgical anatomy during laparoscopic cholecystectomy. *Ann Surg.* (2022) 276(2):363–9. doi: 10.1097/SLA.00000000004594

40. Alkhamaiseh KN, Grantner JL, Shebrain S, Abdel-Qader I. Towards reliable hepatocytic anatomy segmentation in laparoscopic cholecystectomy using u-net with auto-encoder. *Surg Endosc.* (2023) 37(9):7358–69. doi: 10.1007/s00464-023-10306-4

41. Loukas C, Sgouros NP. Multi-instance multi-label learning for surgical image annotation. Int J Med Rob Comput Assist Surg. (2020) 16(2):e2058. doi: 10.1002/rcs. 2058

42. Monasterio-Exposito L, Pizarro D, Macias-Guarasa J. Label augmentation to improve generalization of deep learning semantic segmentation of laparoscopic images. *IEEE Access.* (2022) 10:37345–59. doi: 10.1109/ACCESS.2022.3162630

43. Batić D, Holm F, Özsoy E, Czempiel T, Navab N. Whether and when does endoscopy domain pretraining make sense? *arXiv* [Preprint]. *arXiv:2303.17636* (2023).

44. Hong W-Y, Kao C-L, Kuo Y-H, Wang J-R, Chang W-L, Shih C-S. Cholecseg8k: a semantic segmentation dataset for laparoscopic cholecystectomy based on cholec80. *arXiv* [Preprint]. *arXiv:2012.12453* (2020).

45. Murali A, Alapatt D, Mascagni P, Vardazaryan A, Garcia A, Okamoto N, et al. The endoscapes dataset for surgical scene segmentation, object detection, and critical view of safety assessment: official splits and benchmark. *arXiv* [Preprint]. *arXiv:2312.12429* (2023).

46. Ríos MS, Molina-Rodriguez MA, Londoño D, Guillén CA, Sierra S, Zapata F, et al. Cholec80-CVS: an open dataset with an evaluation of strasberg's critical view of safety for AI. *Sci Data*. (2023) 10(1):194. doi: 10.1038/s41597-023-02073-7

47. Leibetseder A, Kletz S, Schoeffmann K, Keckstein S, Keckstein J. Glenda: gynecologic laparoscopy endometriosis dataset. In: *International Conference on Multimedia Modeling*. Springer (2019). p. 439–50.

48. Hayashi Y, Misawa K, Mori K. Surgical area recognition from laparoscopic images in laparoscopic gastrectomy for gastric cancer using label smoothing and uncertainty. In: *Medical Imaging 2023: Image-Guided Procedures, Robotic Interventions, and Modeling.* SPIE (2023). Vol. 12466. p. 508–15.

49. Mascagni P, Vardazaryan A, Alapatt D, Urade T, Emre T, Fiorillo C, et al. Artificial intelligence for surgical safety: automatic assessment of the critical view of safety in laparoscopic cholecystectomy using deep learning. *Ann Surg.* (2022) 275(5):955–61. doi: 10.1097/SLA.00000000004351

50. Murali A, Alapatt D, Mascagni P, Vardazaryan A, Garcia A, Okamoto N, et al. Encoding surgical videos as latent spatiotemporal graphs for object and anatomydriven reasoning. In: *International Conference on Medical Image Computing and Computer-Assisted Intervention*. Springer (2023). p. 647–57.

51. Chang Y, Wang X, Wang J, Wu Y, Yang L, Zhu K, et al. A survey on evaluation of large language models. *ACM Trans Intell Syst Technol.* (2024) 15(3):1–45. doi: 10. 1145/3641289

52. Zhang J, Huang J, Jin S, Lu S. Vision-language models for vision tasks: a survey. *IEEE Trans Pattern Anal Mach Intell.* (2024) 46(8):5625–44. doi: 10.1109/TPAMI. 2024.3369699

53. Bazi Y, Mahmoud Al Rahhal M, Bashmal L, Zuair M. Vision-language model for visual question answering in medical imagery. *Bioengineering*. (2023) 10(3):380. doi: 10.3390/bioengineering10030380

54. Hirides S, Hirides P, Kalliopi K, Hirides C. Artificial intelligence and computer vision during surgery: discussing laparoscopic images with chatgpt4—preliminary results. *Surg Sci.* (2024) 15(3):169–81. doi: 10.4236/ss.2024.153017

55. Seenivasan L, Islam M, Kannan G, Ren H. Surgicalgpt: end-to-end language-vision gpt for visual question answering in surgery. In: *International Conference on Medical Image Computing and Computer-Assisted Intervention*. Springer (2023). p. 281–90.

56. Xi N, Meng J, Yuan J. Chain-of-look prompting for verb-centric surgical triplet recognition in endoscopic videos. In: *Proceedings of the 31st ACM International Conference on Multimedia.* (2023). p. 5007–16.

57. Alabi O, Vercauteren T, Shi M. Multitask learning in minimally invasive surgical vision: a review. arXiv [Preprint]. arXiv:2401.08256 (2024).

58. Nwoye CI, Yu T, Gonzalez C, Seeliger B, Mascagni P, Mutter D, et al. Rendezvous: attention mechanisms for the recognition of surgical action triplets in endoscopic videos. *Med Image Anal.* (2022) 78:102433. doi: 10.1016/j.media.2022.102433

59. Li Y, Xia T, Luo H, He B, Jia F. MT-FiST: a multi-task fine-grained spatialtemporal framework for surgical action triplet recognition. *IEEE J Biomed Health Inform.* (2023) 27(10):4983–94. doi: 10.1109/JBHI.2023.3299321

60. Yamlahi A, Nuong Tran T, Godau P, Schellenberg M, Michael D, Smidt F-H, et al. Self-distillation for surgical action recognition. In: *International Conference on Medical Image Computing and Computer-Assisted Intervention*. Springer (2023). p. 637–46.

61. Sharma S, Innocent Nwoye C, Mutter D, Padoy N. Surgical action triplet detection by mixed supervised learning of instrument-tissue interactions. In: *International Conference on Medical Image Computing and Computer-Assisted Intervention.* Springer (2023). p. 505–14.

62. Igaki T, Kitaguchi D, Kojima S, Hasegawa H, Takeshita N, Mori K, et al. Artificial intelligence-based total mesorectal excision plane navigation in laparoscopic colorectal surgery. *Dis Colon Rectum.* (2022) 65(5):e329–33. doi: 10. 1097/DCR.00000000002393

63. Wang Z, Lu B, Long Y, Zhong F, Cheung T-H, Dou Q, et al. Autolaparo: a new dataset of integrated multi-tasks for image-guided surgical automation in laparoscopic hysterectomy. In: *International Conference on Medical Image Computing and Computer-Assisted Intervention.* Springer (2022). p. 486–96.

64. Ronneberger O, Fischer P, Brox T. U-net: convolutional networks for biomedical image segmentation. In: Medical Image Computing and Computer-Assisted Intervention-MICCAI 2015: 18th International Conference, Munich, Germany, October 5-9, 2015, Proceedings, Part III 18. Springer (2015). p. 234–41.

65. Zadeh SM, François T, Comptour A, Canis M, Bourdel N, Bartoli A. Surgai3. 8k: a labeled dataset of gynecologic organs in laparoscopy with application to automatic augmented reality surgical guidance. *J Minim Invasive Gynecol.* (2023) 30(5):397–405. doi: 10.1016/j.jmig.2023.01.012

66. Bardozzo F, Delli Priscoli M, Collins T, Forgione A, Hostettler A, Tagliaferri R. Cross X-AI: explainable semantic segmentation of laparoscopic images in relation to depth estimation. In: 2022 International Joint Conference on Neural Networks (IJCNN). IEEE (2022). p. 1–8.

67. Silva B, Oliveira B, Morais P, Buschle LR, Correia-Pinto J, Lima E, et al. Analysis of current deep learning networks for semantic segmentation of anatomical structures in laparoscopic surgery. In: 2022 44th Annual International Conference of the IEEE Engineering in Medicine & Biology Society (EMBC). IEEE (2022). p. 3502–5.

68. Hattab G, Arnold M, Strenger L, Allan M, Arsentjeva D, Gold O, et al. Kidney edge detection in laparoscopic image data for computer-assisted surgery: kidney edge detection. *Int J Comput Assist Radiol Surg.* (2020) 15:379–87. doi: 10.1007/s11548-019-02102-0

69. François T, Calvet L, Madad Zadeh S, Saboul D, Gasparini S, Samarakoon P, et al. Detecting the occluding contours of the uterus to automatise augmented laparoscopy: score, loss, dataset, evaluation and user study. *Int J Comput Assist Radiol Surg.* (2020) 15:1177–86. doi: 10.1007/s11548-020-02151-w

70. Casella A, Moccia S, Carlini C, Frontoni E, De Momi E, Mattos LS. Nephcnn: a deep-learning framework for vessel segmentation in nephrectomy laparoscopic videos. In: 2020 25th International Conference on Pattern Recognition (ICPR). IEEE (2021). p. 6144–9.

71. Uramoto L, Hayashi Y, Oda M, Kitasaka T, Misawa K, Mori K. A semantic segmentation method for laparoscopic images using semantically similar groups. In: *Medical Imaging 2023: Image-Guided Procedures, Robotic Interventions, and Modeling.* SPIE (2023). Vol. 12466. p. 24–31.

72. Khalid MU, Laplante S, Masino C, Alseidi A, Jayaraman S, Zhang H, et al. Use of artificial intelligence for decision-support to avoid high-risk behaviors during laparoscopic cholecystectomy. *Surg Endosc.* (2023) 37(12):9467–75. doi: 10.1007/s00464-023-10403-4

73. Laplante S, Namazi B, Kiani P, Hashimoto DA, Alseidi A, Pasten M, et al. Validation of an artificial intelligence platform for the guidance of safe laparoscopic cholecystectomy. *Surg Endosc.* (2023) 37(3):2260–8. doi: 10.1007/s00464-022-09439-9

74. Fu Y, Robu MR, Koo B, Schneider C, van Laarhoven S, Stoyanov D, et al. More unlabelled data or label more data? a study on semi-supervised laparoscopic image segmentation. In: Domain Adaptation and Representation Transfer and Medical Image Learning with Less Labels and Imperfect Data: First MICCAI Workshop, DART 2019, and First International Workshop, MIL3ID 2019, Shenzhen, Held in Conjunction with MICCAI 2019, Shenzhen, China, October 13 and 17, 2019, Proceedings 1. Springer (2019). p. 173–80.

75. Maqbool S, Riaz A, Sajid H, Hasan O. m2caiseg: semantic segmentation of laparoscopic images using convolutional neural networks. *arXiv* [Preprint]. *arXiv:2008.10134* (2020).

76. Qiu J, Hayashi Y, Oda M, Kitasaka T, Mori K. Class-wise confidence-aware active learning for laparoscopic images segmentation. *Int J Comput Assist Radiol Surg.* (2023) 18(3):473-82. doi: 10.1007/s11548-022-02773-2

77. Aklilu J, Yeung S. Alges: active learning with gradient embeddings for semantic segmentation of laparoscopic surgical images. In: *Machine Learning for Healthcare Conference*. PMLR (2022). p. 892–911.

78. Jenke AC, Bodenstedt S, Kolbinger FR, Distler M, Weitz J, Speidel S. One model to use them all: training a segmentation model with complementary datasets. *arXiv* [Preprint]. *arXiv:2402.19340* (2024).

79. Zhang L, Hayashi Y, Oda M, Mori K. Towards better laparoscopic video segmentation: a class-wise contrastive learning approach with multi-scale feature extraction. *Healthc Technol Lett.* (2024) 11(2–3):126–36. doi: 10.1049/htl2.12069

80. Fuentes-Hurtado F, Kadkhodamohammadi A, Flouty E, Barbarisi S, Luengo I, Stoyanov D. Easylabels: weak labels for scene segmentation in laparoscopic videos. *Int J Comput Assist Radiol Surg.* (2019) 14:1247–57. doi: 10.1007/s11548-019-02003-2

81. Zhao X, Hayashi Y, Oda M, Kitasaka T, Mori K. Masked frequency consistency for domain-adaptive semantic segmentation of laparoscopic images. In: *International Conference on Medical Image Computing and Computer-Assisted Intervention*. Springer (2023). p. 663–73.

82. Owen D, Grammatikopoulou M, Luengo I, Stoyanov D. Detection of critical structures in laparoscopic cholecystectomy using label relaxation and self-supervision. In: *Medical Image Computing and Computer Assisted Intervention-MICCAI 2021: 24th International Conference, Strasbourg, France, September 27-October 1, 2021, Proceedings, Part IV 24.* Springer (2021). p. 321–30.

83. Owen D, Grammatikopoulou M, Luengo I, Stoyanov D. Automated identification of critical structures in laparoscopic cholecystectomy. *Int J Comput Assist Radiol Surg.* (2022) 17(12):2173–81. doi: 10.1007/s11548-022-02771-4

84. Ramesh S, Srivastav V, Alapatt D, Yu T, Murali A, Sestini L, et al. Dissecting selfsupervised learning methods for surgical computer vision. *Med Image Anal.* (2023) 88:102844. doi: 10.1016/j.media.2023.102844

85. Zhuang F, Qi Z, Duan K, Xi D, Zhu Y, Zhu H, et al. A comprehensive survey on transfer learning. *Proc IEEE*. (2020) 109(1):43–76. doi: 10.1109/JPROC.2020.3004555

86. Kitaguchi D, Takeshita N, Matsuzaki H, Igaki T, Hasegawa H, Kojima S, et al. Realtime vascular anatomical image navigation for laparoscopic surgery: experimental study. *Surg Endosc.* (2022) 36(8):6105–12. doi: 10.1007/s00464-022-09384-7

87. Yoon J, Hong SG, Hong S, Lee J, Shin S, Park B, et al. Surgical scene segmentation using semantic image synthesis with a virtual surgery environment. In: *International Conference on Medical Image Computing and Computer-Assisted Intervention*. Springer (2022). p. 551–61.

88. Une N, Kobayashi S, Kitaguchi D, Sunakawa T, Sasaki K, Ogane T, et al. Intraoperative artificial intelligence system identifying liver vessels in laparoscopic liver resection: a retrospective experimental study. *Surg Endosc.* (2024) 38 (2):1088-95. doi: 10.1007/s00464-023-10637-2

89. Vaswani A, Shazeer N, Parmar N, Uszkoreit J, Jones L, Gomez AN, et al. Attention is all you need. *Adv Neural Inf Process Syst.* (2017) 30:36–47. doi: 10. 48550/arXiv.1706.03762

90. Maack L, Behrendt F, Bhattacharya D, Latus S, Schlaefer A. Efficient anatomy segmentation in laparoscopic surgery using multi-teacher knowledge distillation. In: *Medical Imaging with Deep Learning.* (2024).

91. Prokopetc K, Collins T, Bartoli A. Automatic detection of the uterus and fallopian tube junctions in laparoscopic images. In: *Information Processing in Medical Imaging: 24th International Conference, IPMI 2015, Sabhal Mor Ostaig, Isle of Skye, UK, June 28-July 3, 2015, Proceedings 24. Springer (2015).* p. 552-63.

92. Nakasuji H, Naruki K, Kawai T, Nishikawa A, Nishizawa Y, Nakamura T. Image recognition of triangular tissue of an organ pulled by forceps in surgical working area for laparoscope robot. In: 2017 39th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC). IEEE (2017). p. 3708-11.

93. Visalaxi S, Punnoose D, Sudalai Muthu T. Lesion extraction of endometriotic images using open computer vision. In: 2021 International Conference on Artificial Intelligence and Smart Systems (ICAIS). IEEE (2021). p. 747-51.

94. Fujinaga A, Endo Y, Etoh T, Kawamura M, Nakanuma H, Kawasaki T, et al. Development of a cross-artificial intelligence system for identifying intraoperative anatomical landmarks and surgical phases during laparoscopic cholecystectomy. *Surg Endosc.* (2023) 37(8):6118–28. doi: 10.1007/s00464-023-10097-8

95. Endo Y, Tokuyasu T, Mori Y, Asai K, Umezawa A, Kawamura M, et al. Impact of AI system on recognition for anatomical landmarks related to reducing bile duct injury during laparoscopic cholecystectomy. *Surg Endosc.* (2023) 37(7):5752–9. doi: 10.1007/s00464-023-10224-5

96. Loukas CG, Schizas ID. Assessment of gallbladder wall vascularity from laparoscopic images using deep learning. In: *BIOIMAGING*. (2020). p. 28–36.

97. Mascagni P, Alapatt D, Lapergola A, Vardazaryan A, Mazellier J-P, Dallemagne B, et al. Early-stage clinical evaluation of real-time artificial intelligence assistance for laparoscopic cholecystectomy. *Br J Surg.* (2024) 111(1):znad353. doi: 10.1093/bjs/znad353

98. Yang Z, Wen J, Huang D, Yang A, Zhang R, Ren B, et al. Efficient segmentation of intraoperative anatomical landmarks in laparoscopic cholecystectomy based on deep learning. Available online at: SSRN4760261 (2024).

99. Moens K, De Vylder J, Blaschko MB, Tuytelaars T. Laparoflow-SSL: image analysis from a tiny dataset through self-supervised transformers leveraging unlabeled surgical video. In: *Proceedings of Machine Learning Research, Paris, France.* MIDL (2024) Vol. 250, p. 0–25..

100. Leibetseder A, Petscharnig S, Jürgen Primus M, Kletz S, Münzer B, Schoeffmann K, et al. Lapgyn4: a dataset for 4 automatic content analysis problems in the domain of laparoscopic gynecology. In: *Proceedings of the 9th ACM Multimedia Systems Conference.* (2018). p. 357–62.

101. Ghamsarian N, Wolf S, Zinkernagel M, Schoeffmann K, Sznitman R. Deeppyramid+: medical image segmentation using pyramid view fusion and deformable pyramid reception. *Int J Comput Assist Radiol Surg.* (2024) 19(5):851–9. doi: 10.1007/s11548-023-03046-2

102. Scheikl PM, Laschewski S, Kisilenko A, Davitashvili T, Müller B, Capek M, et al. Deep learning for semantic segmentation of organs and tissues in laparoscopic surgery. *Curr Dir Biomed Eng.* (2020) 6:20200016. doi: 10.1515/cdbme-2020-0016

103. Colleoni E, Sanchez Matilla R, Luengo I, Stoyanov D. Guided image generation for improved surgical image segmentation. *Med Image Anal.* (2024) 97:103263. doi: 10.1016/j.media.2024.103263

104. Wang J, Tang H, Kantor T, Soltani T, Popov V, Wang X. Surgment: segmentation-enabled semantic search and creation of visual question and feedback to support video-based surgery learning. In: *Proceedings of the CHI Conference on Human Factors in Computing Systems*. (2024). p. 1–18.

105. Urrea C, Garcia-Garcia Y, Kern J. Improving surgical scene semantic segmentation through a deep learning architecture with attention to class imbalance. *Biomedicines*. (2024) 12(6):1309. doi: 10.3390/biomedicines12061309

106. Grammatikopoulou M, Sanchez-Matilla R, Bragman F, Owen D, Culshaw L, Kerr K, et al. A spatio-temporal network for video semantic segmentation in surgical videos. *Int J Comput Assist Radiol Surg.* (2024) 19(2):375–82. doi: 10.1007/s11548-023-02971-6

107. Sharma S, Innocent Nwoye C, Mutter D, Padoy N. Rendezvous in time: an attention-based temporal fusion approach for surgical triplet recognition. *Int J Comput Assist Radiol Surg.* (2023) 18(6):1053–9. doi: 10.1007/s11548-023-02914-1

108. Kolbinger FR, He J, Ma J, Zhu F. Strategies to improve real-world applicability of laparoscopic anatomy segmentation models. In: *Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition*. (2024). p. 2275–84.

109. Ward TM, Hashimoto DA, Ban Y, Rosman G, Meireles OR. Artificial intelligence prediction of cholecystectomy operative course from automated identification of gallbladder inflammation. *Surg Endosc.* (2022) 36(9):6832–40. doi: 10.1007/s00464-022-09009-z

110. Mascagni P, Alapatt D, Urade T, Vardazaryan A, Mutter D, Marescaux J, et al. A computer vision platform to automatically locate critical events in surgical videos: documenting safety in laparoscopic cholecystectomy. *Ann Surg.* (2021) 274(1): e93–5. doi: 10.1097/SLA.00000000004736

111. He Q, Bano S, Ahmad OF, Yang B, Chen X, Valdastri P, et al. Deep learningbased anatomical site classification for upper gastrointestinal endoscopy. *Int J Comput Assist Radiol Surg.* (2020) 15:1085–94. doi: 10.1007/s11548-020-02148-5

112. Gerkema MH. Deep learning for identification of gallbladder leakage during laparoscopic cholecystectomy (Master's thesis). University of Twente, Twente (2020).

113. Moccia S, Wirkert SJ, Kenngott H, Vemuri AS, Apitz M, Mayer B, et al. Uncertainty-aware organ classification for surgical data science applications in laparoscopy. *IEEE Trans Biomed Eng.* (2018) 65(11):2649–59. doi: 10.1109/TBME. 2018.2813015

114. Harangi B, Hajdu A, Lampe R, Torok P. Recognizing ureter and uterine artery in endoscopic images using a convolutional neural network. In: 2017 IEEE 30th International Symposium on Computer-Based Medical Systems (CBMS). IEEE (2017). p. 726–7.

115. Zhang Y, Wirkert SJ, Iszatt J, Kenngott H, Wagner M, Mayer B, et al. Tissue classification for laparoscopic image understanding based on multispectral texture analysis. *J Med Imaging.* (2017) 4(1):015001. doi: 10.1117/1.JMI.4.1.015001

116. Serban N, Kupas D, Hajdu A, Török P, Harangi B. Distinguishing the uterine artery, the ureter, and nerves in laparoscopic surgical images using ensembles of binary semantic segmentation networks. *Sensors*. (2024) 24(9):2926. doi: 10.3390/s24092926

117. Park J-H, Jae Kim Y, Kim KG. Semantic segmentation networks of organs in minimally invasive surgery. *J Innov Med Technol.* (2024) 2(1):11–9. doi: 10.61940/jimt.240002

118. Tseng L-A, Lin H-C, Bai M-Y, Li M-F, Lee Y-L, Chiang K-J, et al. Deepvinci: a semantic segmentation model with edge super-vision and densely multi-scale pyramid module for davinci gynecological surgery. (2024).

119. Pei J, Cui R, Li Y, Si W, Qin J, Heng P-A. Depth-driven geometric prompt learning for laparoscopic liver landmark detection. *arXiv* [Preprint]. *arXiv:2406.17858* (2024).

120. Ryu K, Kitaguchi D, Nakajima K, Ishikawa Y, Harai Y, Yamada A, et al. Deep learning-based vessel automatic recognition for laparoscopic right hemicolectomy. *Surg Endosc.* (2024) 38(1):171–8. doi: 10.1007/s00464-023-10524-w

121. Narihiro S, Kitaguchi D, Hasegawa H, Takeshita N, Ito M. Deep learning-based real-time ureter identification in laparoscopic colorectal surgery. *Dis Colon Rectum.* (2024) 67(10):e1596–9. doi: 10.1097/DCR.00000000003335.

122. Mehta P, Owen D, Grammatikopoulou M, Culshaw L, Kerr K, Stoyanov D, et al. Hierarchical segmentation of surgical scenes in laparoscopy. *Int J Comput Assist Radiol Surg.* (2024) 19(7):1449–57. doi: 10.1007/s11548-024-03157-4

123. Wang Z, Liu C, Zhang Z, Deng Y, Xiao M, Zhang Z, et al. Real-time autosegmentation of the ureter in video sequences of gynaecological laparoscopic surgery. Int J Med Rob Comput Assist Surg. (2024) 20(1):e2604. doi: 10.1002/rcs.2604

124. Sengun B, Iscan Y, Ozbulak GAT, Kumbasar N, Egriboz E, Sormaz IC, et al. Artificial intelligence in minimally invasive adrenalectomy: using deep learning to identify the left adrenal vein. *Surg Laparosc Endosc Percutan Tech.* (2023) 33(4):327–31. doi: 10.1097/SLE.000000000001185

125. Zhang X, Ji X, Wang J, Fan Y, Tao C. Renal surface reconstruction and segmentation for image-guided surgical navigation of laparoscopic partial nephrectomy. *Biomed Eng Lett.* (2023) 13(2):165–74. doi: 10.1007/s13534-023-00263-1

126. Tashiro Y, Aoki T, Kobayashi N, Tomioka K, Kumazu Y, Akabane M, et al. Color-coded laparoscopic liver resection using artificial intelligence: a preliminary study. J Hepatobiliary Pancreat Sci. (2024) 31:67–8. doi: 10.1002/jhbp.1388

127. Tomioka K, Aoki T, Kobayashi N, Tashiro Y, Kumazu Y, Shibata H, et al. Development of a novel artificial intelligence system for laparoscopic hepatectomy. *Anticancer Res.* (2023) 43(11):5235–43. doi: 10.21873/anticanres.16725

128. Kojima S, Kitaguchi D, Igaki T, Nakajima K, Ishikawa Y, Harai Y, et al. Deeplearning-based semantic segmentation of autonomic nerves from laparoscopic images of colorectal surgery: an experimental pilot study. *Int J Surg.* (2023) 109(4):813–20. doi: 10.1097/JS9.00000000000317

129. Souza CA, Menegatti JE, Pazello RT, Hajar F. Neural network image segmentation model for laparoscopic gynecological surgeries. *J Minim Invasive Gynecol.* (2021) 28(11):S18. doi: 10.1016/j.jmig.2021.09.051

130. Bamba Y, Ogawa S, Itabashi M, Shindo H, Kameoka S, Okamoto T, et al. Object and anatomical feature recognition in surgical video images based on a convolutional neural network. *Int J Comput Assist Radiol Surg.* (2021) 16(11):2045–54. doi: 10.1007/s11548-021-02434-w

131. Altieri M, Hashimoto D, María Rivera A, Namazi B, Alseidi A, Okrainec A, et al. Using artificial intelligence to identify surgical anatomy, safe zones of dissection, and dangerous zones of dissection during laparoscopic cholecystectomy. *J Am Coll Surg.* (2020) 231(4):e21–2. doi: 10.1016/j.jamcollsurg.2020.08.054

132. Fraile AN. Implementation of a tracking and segmentation algorithm of structures in laparoscopic surgery video (Ph.D thesis). Telecomunicacion, Madrid, Spain (2019).

133. Manabe T, Tomonaga K, Shibata Y. Cnn architecture for surgical image segmentation systems with recursive network structure to mitigate overfitting. In: 2019 Seventh International Symposium on Computing and Networking (CANDAR). IEEE (2019). p. 171-7.

134. Kadkhodamohammadi A, Luengo I, Barbarisi S, Taleb H, Flouty E, Stoyanov D. Feature aggregation decoder for segmenting laparoscopic scenes. In: *International Workshop on OR 2.0 Context-Aware Operating Theaters*. Springer (2019). p. 3–11.

135. Gibson E, Robu MR, Thompson S, Eddie Edwards P, Schneider C, Gurusamy K, et al. Deep residual networks for automatic segmentation of laparoscopic videos of the liver. In: *Medical Imaging 2017: Image-Guided Procedures, Robotic Interventions, and Modeling.* SPIE (2017). Vol. 10135. p. 423–8.

136. Bodenstedt S, Görtler J, Wagner M, Kenngott H, Müller-Stich BP, Dillmann R, et al. Superpixel-based structure classification for laparoscopic surgery. In: *Medical Imaging 2016: Image-Guided Procedures, Robotic Interventions, and Modeling.* SPIE (2016). Vol. 9786. p. 386–91.

137. Haouchine N, Cotin S. Segmentation and labelling of intra-operative laparoscopic images using structure from point cloud. In: 2016 IEEE 13th International Symposium on Biomedical Imaging (ISBI). IEEE (2016). p. 115–8.

138. Yu Z, Liu J, Ding H, Lu M, Deng M, Huang Y, et al. An image-based method of uterus segmentation in gynecologic laparoscopy. *J Med Imaging Health Inform.* (2015) 5(4):819–25. doi: 10.1166/jmihi.2015.1463

139. Collins T, Bartoli A, Bourdel N, Canis M. Segmenting the uterus in monocular laparoscopic images without manual input. In: *Medical Image Computing and Computer-Assisted Intervention–MICCAI 2015: 18th International Conference, Munich, Germany, October 5–9, 2015, Proceedings, Part III 18.* Springer (2015). p. 181–9.

140. Chhatkuli A, Bartoli A, Malti A, Collins T. Live image parsing in uterine laparoscopy. In: 2014 IEEE 11th International Symposium on Biomedical Imaging (ISBI). IEEE (2014). p. 1263-6.

141. Smithmaitrie P, Khaonualsri M, Sae-Lim W, Wangkulangkul P, Jearanai S, Cheewatanakornkul S. Development of deep learning framework for anatomical landmark detection and guided dissection line during laparoscopic cholecystectomy. *Heliyon.* (2024) 10(3):e25210. doi: 10.1016/j. heliyon.2024.e25210

142. Zygomalas A, Kalles D, Katsiakis N, Anastasopoulos A, Skroubis G. Artificial intelligence assisted recognition of anatomical landmarks and laparoscopic instruments in transabdominal preperitoneal inguinal hernia repair. *Surg Innov.* (2024) 31(2):178–84. doi: 10.1177/15533506241226502

143. Liu R, An J, Wang Z, Guan J, Liu J, Jiang J, et al. Artificial intelligence in laparoscopic cholecystectomy: does computer vision outperform human vision. *Artif Intell Surg.* (2022) 2(2):80–92. doi: 10.20517/ais.2022.04

144. Tokuyasu T, Iwashita Y, Matsunobu Y, Kamiyama T, Ishikake M, Sakaguchi S, et al. Development of an artificial intelligence system using deep learning to indicate anatomical landmarks during laparoscopic cholecystectomy. *Surg Endosc.* (2021) 35:1651–8. doi: 10.1007/s00464-020-07548-x

145. Pozdeev AA, Obukhova NA, Motyko AA. Anatomical landmarks detection for laparoscopic surgery based on deep learning technology. In: 2021 IEEE Conference of Russian Young Researchers in Electrical and Electronic Engineering (ElConRus). IEEE (2021). p. 1668–72.